COLUMBIA SOUTHERN UNIVERSITY

The Relationship Between Transformational Leadership, Organizational Safety Climate, and Job

Satisfaction: A Quantitative Study

by

Amy Lynn Koehler-Warren

A Dissertation Presented in Partial Fulfillment

of the Requirements for the Degree

Doctorate of Business Administration

July 26, 2023

APPROVED:

× Janny & Corcoran PhD

Dr. Tammy Corcoran, Ph.D., Dissertation Chair

Sucker, PhD Januce P.

Dr. Jan Tucker, Ph.D., Committee Member

ACCEPTED AND SIGNED:

Dr. Darlene Jaffke, DBA Academic Program Director DBA

Dr. Darlene Jafflac

Dr. Elwin Jones, Ph.D, MBA Dean for the College of Business Assistant Provost for Academic Affairs

lon

© by Amy Lynn Koehler-Warren, 2023

All rights reserved.

ABSTRACT

The Occupational Safety and Health Act of 1970 resulted in the establishment of the Occupational Safety and Health Administration, which made workplaces safer for employees. Although obvious hazards have decreased in many American workplaces, a supportive environment for occupational safety is something that organizational leaders must cultivate. Some leaders place a greater emphasis on workplace welfare creation because they are much more concerned with the welfare of their workers. The problem that will be examined in this study is that many organizations ignore safety culture altogether, which can result in diminished job satisfaction. The research question for this study is what is the relationship between transformational leadership, occupational safety climate, and job satisfaction of oil drilling and gas extraction industry workers in the United States? This study utilizes quantitative methodology. The research design utilized for this study is a non-experimental correlational approach that allows the researchers to examine the direction and strength of relationships and interactions between the study's variables. The researcher used Centiment to recruit 107 participants for this study who self-identified as adults that work in the oil drilling and gas extraction industry to complete an online survey. The theoretical foundations for this study were Bernard Bass's transformational leadership theory, H. W. Heinrich's domino theory, and Abraham Maslow's hierarchy of needs theory. Correlational tests were performed, and the findings showed a statistically significant relationship between the variables.

Keywords: transformational leadership, occupational safety climate, job satisfaction, oil drilling, gas extraction.

DEDICATION

I would like to dedicate this dissertation to my dad, Joe R. Huff Sr. I have spent a lifetime trying to impress him and make him proud. He is one of the smartest, wisest men in this world. At the beginning of this journey, it was expressed to me that getting a doctoral degree wasn't important in his opinion. I think that his exact words were "I don't care one way or the other because you won't be my doctor." Well, now as your daughter and Doctor of Business Administration, I am officially your doctor. Love you dad!

ACKNOWLEDGMENTS

I want to sincerely thank Dr. Corcoran for serving as the chair of my dissertation. Her insight, direction, and encouragement from the beginning had a direct impact on my capacity to complete this dissertation successfully. Furthermore, Dr. Jeffrey Hughes, a friend and former DBA candidate, has also served as a source of inspiration. His words of motivation, inspiration, and encouragement got me through the most challenging stretches of this journey and was the one who pushed me to get this done. To my mom, Mrs. Terri Elain Huff, thank you for your love and support in addition to all the words of encouragement along each milestone of the way to the finish line. To my best friend and husband Donnie Gene Warren I would like to express my gratitude. I can never thank God enough for blessing me with your existence in my life.

TABLE OF	CONTENTS
-----------------	-----------------

ABSTRACT iii
LIST OF TABLESix
LIST OF FIGURESx
CHAPTER 1: INTRODUCTION TO THE STUDY1
Background of the Study2
Problem Statement
Purpose Statement
Research Questions and Hypotheses
Advancing Theoretical Knowledge7
Application to Business Administration
Significance of the Study
Methodology 10
Research Design
Definitions of Key Terms
Assumptions, Limitations, and Delimitations
Assumptions 12
Limitations12
Delimitations
Summary and Organization of the Remainder of the Study
CHAPTER 2: LITERATURE REVIEW

Theoretical Foundations	18
Review of the Literature	24
Transformational Leadership	25
Instrumentation	37
Summary	39
CHAPTER 3: METHODOLOGY	40
Statement of the Problem	40
Research Questions and Hypotheses	41
Research Methodology	42
Research Design	43
Population and Sample Selection	43
Instrumentation	44
Validity and Reliability	46
Data Collection Procedures	50
Analytical Methods	51
Ethical Considerations	52
Summary	54
CHAPTER 4: DATA ANALYSIS AND RESULTS	55
Data Analysis Procedures	55
Descriptive Data Analysis	56

Results	59
Summary	65
CHAPTER 5: SUMMARY, CONCLUSIONS, AND RECOMMENDATIONS	67
Summary of the Study	68
Summary of Findings and Conclusion	68
Implications	69
Theoretical Implications	69
Implications for Professional Practice	70
Recommendations	71
Recommendations for Future Research	72
Recommendations for Practice	73
Concluding Remarks	74
REFERENCES	76
APPENDIX A	90
APPENDIX B	92

LIST OF TABLES

Table 1: Intercorrelations Among MLQ 5x-short Factor Scores	47
Table 2: Content and Reliability of the Seven Safety Climate Scales	49
Table 3: Internal Consistency, Split-Half, and Test-Retest Reliability of JSS	50
Table 4: Demographic Statistics From Survey Data Set	58
Table 5: Demographic Statistics From Survey Data Set	58
Table 6: Spearman Rho Correlation Coefficients for the MLQ 5x-short, NOSACQ-50), & JSS 59

LIST OF FIGURES

Figure 1: Bernard Bass's Transformational Leadership Model	20
Figure 2: H. W. Heinrich's Safety Triangle	22
Figure 3: Abraham Maslow's Hierarchy of Needs Pyramid	24
Figure 4: Sample Size Determination Using G*Power	44
Figure 5: Post hoc G*Power Analysis	57

CHAPTER 1: INTRODUCTION TO THE STUDY

Since the Occupational Safety and Health Administration (OSHA, 1970) was created because of the Occupational Safety and Health Act of 1970, workplaces have become safer for employees. Nevertheless, while blatant hazards have diminished in many American workplaces, a positive occupational safety climate does not happen by accident; it is something that needs to be nurtured by organizational leaders. Some leaders are much more focused on the welfare of their employees and, as a result, will place greater importance on workplace welfare creation.

The purpose of this study was to examine the relationship between transformational leadership, occupational safety climate, and job satisfaction among US oil drilling and gas extraction industry workers. If it can be determined that transformational leadership and a positive occupational safety climate contribute to increased job satisfaction, organizations may see the value in conducting transformational leadership training and focus on safety climate to increase job satisfaction, reduce mishaps in the workplace, reduce mishap costs, and increase profit and productivity (Sánchez-Segura et al., 2021).

This research differs from other research on transformational leadership because the effect of occupational safety climate on job satisfaction was also examined. This study extends prior research on the topic by providing information to business leaders on the impact transformational leadership style has on the organization's safety climate and job satisfaction, which can potentially improve the organization. Chapter 1 includes an overview of the study that explains the research problem, a brief history and current state of the research problem, the purpose of the study, the significance of the study, and the research design and methodology.

Background of the Study

Leadership style is defined as techniques used to direct and influence the activities in an organization (Asgari et al., 2020). Transformational leadership style is a commonly studied concept (Akdere & Egan, 2020; Alimo-Metcalfe & Alban-Metcalfe, 2001; Kines et al, 2011; Loh et al., 2021). However, what is not widely studied is the impact transformational leadership and occupational safety climate have on job satisfaction in US oil drilling and gas extraction industry workers. This study aimed to contribute to the literature in four ways: First, this study assessed the impact of transformational leadership on occupational safety climate. Second, this study added to the transformational leadership theory by exploring how the elements of occupational safety climate are affected by transformational leadership. Third, this study investigated how transformational leadership affects job satisfaction. Fourth, this study examined how an employee's perception of an organization's occupational safety climate affects job satisfaction. Examining the role of transformational leadership and measuring the perception of occupational safety climate to determine how they affect job satisfaction also introduces new constructs to the literature.

This study presents the state of research in the fields of transformational leadership, occupational safety climate, and job satisfaction. Separately, these variables can influence an organization. The significance of this study is that it sheds light on possible effects of the relationship between these variables for business leaders in the oil and gas extraction industry. Finally, oil and gas extraction industry leaders will gain the understanding of how the relationship between these variables can affect their organizations.

Problem Statement

The problem that was examined in this study is that many organizations ignore safety culture altogether, which can result in diminished job satisfaction (Niartiningsih et al., 2020). Reasons for this may include the lack of importance leaders place on safety climate, few efforts to improve safety culture, and ignorance. According to Lal (2022), most organizations will only make concrete health and safety decisions once a loss is experienced. This type of decision would be an example of ignoring safety as part of the organization's safety culture until changes have to be implemented. Safety behavior and performance in many industries are impacted on the safety attitude of its workers and managers. (Marquardt et al., 2020). Furthermore, employees' satisfaction with their jobs and desire to remain in an organization is affected by trust, which is developed by working in an environment where safety is considered important by management and communications surrounding safety are welcomed (Judge, 2018). Therefore, job satisfaction and worker safety should be a concern for business leaders because an organization's safety culture is directly related to job satisfaction (Niartiningsih et al., 2020).

Organizational leaders drive the climate of an organization, including its safety climate (Beus et al., 2017). Research demonstrates that there is a substantial relationship between transformational leadership and safety climate and safety climate is often used as a mediator between transformational leadership and some other variable. For example, Draghici et al. (2022) looked at safety climate as a mediator between transformation leadership and safe behavior. Mirza and Isha (2020) examined how transformational leadership and safety climate impact occupational accidents. Safety-specific transformational leadership has positive outcomes on workplace safety in general. Organizational leaders have the ability and responsibility to create an atmosphere that promotes employees' overall wellness and security. An organization's

safety climate and its employees' perceptions of the safety climate are influenced by the actions and inactions of its leaders.

An exhaustive literature review was conducted, and no empirical studies were identified that document the influence that transformational leadership has on occupational safety climate and how that organizational safety climate affects job satisfaction in the oil drilling and gas extraction industry. This study provides valuable information to leaders that can result in the implementation of strategies that increase production, reduce injuries, promote a safer work environment, and lead to greater job satisfaction. In the past, organizations were static environments that required little change; however, businesses now compete in dynamic environments that are constantly changing, which requires more dynamic leadership representation (Asgari et al., 2020). Organizations can stay competitive in today's marketplace if they stay abreast of leadership research and implement strategies to hire and train effective leaders (Asgari et al., 2020). This research bridges the gap in the literature on how transformational leadership and occupational safety climate can improve job satisfaction. In addition, research shows that transformational leadership has a role in organizational support behavior (Asgari et al., 2020). Therefore, for the purpose of this research, organizational support behavior includes the organization's occupational safety climate. Furthermore, organizational climate is defined as a set of shared perceptions about an organization's procedures, principles, ideals, standards, and practices (Gyekye & Salminen, 2007). Therefore, the occupational safety climate is part of the overall organizational climate.

Research also shows a connection between transformational leadership and job satisfaction (Asgari et al., 2020). When employees perceive transformational leaders as polite, job satisfaction levels are promoted (Akdere & Egan, 2020; McCallaghan et al., 2019). However,

there is a gap in the literature examining the relationship between transformational leadership, occupational safety climate, and job satisfaction among US oil drilling and gas extraction industry workers. A robust occupational safety climate can contribute to an individual's safety needs on Maslow's hierarchy of needs (Osman et al., 2019). A positive relationship between transformational leadership and job satisfaction is evident in the results of a research study by Steinmann et al. (2018). The results of previous studies indicate that transformational leadership can improve organizational safety climate level; however, the literature on the relationship between transformational leadership, organizational climate, and job satisfaction is not comprehensive (Akdere & Egan, 2020; Asgari et al., 2020; Badura et al., 2021; Osman et al., 2019). The relationship between transformational leadership, occupational safety climate, and job satisfaction can impact an organization in various ways. Examples of these impacts include promoting engaging behaviors that yields a return for the organization, high levels of organizational commitment and organizational clitizenship behavior (Tan et al., 2021).

Purpose Statement

The purpose of this quantitative correlational study was to examine the relationship between transformational leadership, occupational safety climate, and job satisfaction of US oil drilling and gas extraction industry workers. The survey was distributed, and data was collected by Centiment. The study participants were working adults who are employed by US oil drilling and gas extraction employers.

The data collection instrument was an online survey that consists of the Multifactor Leadership Questionnaire (MLQ 5x-short), the Nordic Occupational Safety Climate Questionnaire NOSCAQ-50, and the Paul Spector Job Satisfaction Survey (JSS). This research adds knowledge to the field of study about the relationship between occupational safety climate

in organizations and how transformational leadership affects occupational safety climate and job satisfaction.

Research Questions and Hypotheses

The following research question defined the intent of the study.

RQ: What is the relationship between transformational leadership, occupational safety climate, and job satisfaction of oil drilling and gas extraction industry workers in the United States?

H1₀: There is no statistically significant relationship between transformational leadership and job satisfaction of oil drilling and gas extraction industry workers in the United States.

H1_a: There is a statistically significant relationship between transformational leadership and job satisfaction of oil drilling and gas extraction industry workers in the United States.

H2₀: There is no statistically significant relationship between perceived occupational safety climate and job satisfaction of oil drilling and gas extraction industry workers in the United States.

 $H2_a$: There is a statistically significant relationship between perceived occupational safety climate and job satisfaction of oil drilling and gas extraction industry workers in the United States.

H3₀: There is no statistically significant relationship between transformational leadership and perceived occupational safety climate of oil drilling and gas extraction industry workers in the United States.

H3_a: There is a statistically significant relationship between transformational leadership and perceived occupational safety climate of oil drilling and gas extraction industry workers in the United States.

Advancing Theoretical Knowledge

The theories that frame this research study are Bass's transformational leadership theory, Heinrich's domino theory, and Maslow's hierarchy of needs. Bernard Bass founded the transformational leadership theory in 1985. Transformational leadership style is and that a transformational leader inspires followers and can alter or outright change followers' beliefs and attitudes (Barbinta et al., 2017).

H. W. Heinrich founded the domino theory in the 1930 (Yorio & Moore, 2018). Heinrich's domino theory is also crucial to this research study because it explains how safety can lead to job satisfaction. It is a classic theory that suggests that any injury that results from an accident occurs because a person acts unsafely or due to unsafe conditions at the workplace (Osman et al., 2019). The theory also suggests that accidents and injuries are proceeded by a fault in the person or social environment.

Abraham Maslow's hierarchy of needs theory was seen in print form for the first time in 1943, (McCleskey & Ruddell, 2020). Maslow's hierarchy of needs theory is one of the bestknown and most commonly used theories to understand and influence an employee's motivation at work (Fallatah & Syed, 2018). In the hierarchy of needs theory, Maslow identified five universal needs that form the basis for all employee motivation. The identified universal needs are physiological, safety, social, esteem, and self-actualization. Maslow's hierarchy of needs theory continues to be examined, tested, and empirically studied, which validates the theory's assertions and premises. These theories will be discussed more in Chapter 2.

Currently, there is research on transformational leadership and job satisfaction; however, there is a gap in research on the relationship between transformational leadership, occupational safety climate, and job satisfaction in US oil drilling and gas extraction industry workers. This

research sought to fill that gap. This study also advanced theory by examining the relationship between the three variables.

Application to Business Administration

Job satisfaction affects every business because high satisfaction can result in employees staying with an organization, thus decreasing hiring costs, training costs, and lost production. Cost reduction can also be achieved by improving safety climate strength, thus reducing mishap costs to the organization (Flatau-Harrison et al., 2020). Additionally, leaders who are responsible for leadership training, safety policies, and employee retention initiatives have a major impact on a business's ability to increase job satisfaction, reduce accidents, and retain employees. These leaders can benefit from examining the results of this study.

Occupational accidents cause pain and increase costs for society, organizations, and individuals (Kines et al., 2011). Given that both transformational leadership and a safe workplace can result in higher job satisfaction (Steinmann et al., 2018), it is imperative for businesses to develop strategies that ensure employees enjoy a safe and healthy work environment. Leaders, who often act as employee advocates, can ensure that the strategies are implemented. Job satisfaction can be enhanced when employees perceive that an organization cares about their well-being.

Improving the occupational safety climate of an organization is about more than just preventing fines from the Occupational Safety and Health Administration (OSHA) after an accident. This study applies to business administration because increased job satisfaction leads to retention, which decreases turnover costs (Akdere & Egan, 2020; Ozkan et al., 2020). Furthermore, organizations that increase safety standards not only send the message that they care about their employees, thus increasing job satisfaction (Judge, 2018), but these safety

standards also reduce injuries, resulting in lower healthcare and worker compensation costs (Free Workplace Safety Program Reduces Workers' Compensation Costs, 2022).

This study demonstrated the importance of the occupational safety climate to industries across the board. Additionally, this study highlighted key findings, implications, and recommendations for business leaders based on the relationship between transformational leadership, occupational safety climate, and job satisfaction. Business leaders can use the results of this study to develop strategies that encourage followers to enrich their abilities, advancing organizational performance and improving their organizational climate.

Significance of the Study

The study of the relationship between transformational leadership, occupational safety climate, and job satisfaction can have significant implications for business leaders. Expanding the knowledge and literature on theses variables can have significant implications for business leaders and researchers alike. This study provided findings that indicate a positive, relationship between the constructs and expand the existing literature on the topic.

There is limited empirical research available that shows how transformational leadership can affect occupational safety and, subsequently, job satisfaction. Moreover, there is no research that links transformational leadership and occupational safety climate to job satisfaction. Expanding the knowledge for these fields can have significant implications for business leaders by providing valuable information that can result in the implementation of strategies that increase production, reduce injuries, promote a safer work environment, and lead to greater job satisfaction.

Methodology

This study utilized a quantitative methodology. Quantitative research is used to address problems in terms of applying statistical numeric data to research problems with the goal of exploring the relationships between variables without inferring causation (Le Tourneau University, 2022). Qualitative research could have been considered; however, it is not the best method because, rather than using unstructured data collection methods such as observations and interviews, hypothesis testing will be used to determine the strength and direction of the relationship between the variables examined in this study (Roberts & Hyatt, 2018). Because this research focused on examining the relationship between the variables and testing existing theories, a quantitative methodology is most appropriate.

The researcher utilized Centiment.co to recruit a convenience sample of US oil drilling and gas extraction industry workers and distribute a survey to gather the data necessary to test the three hypotheses that were presented earlier in this chapter.

Research Design

The research design utilized for this study is correlational, which allows researchers to examine the direction and strength of relationships between variables (Limberg et al., 2021). Correlational research design allowed the researcher to study constructs that are not directly observable (Limberg et al., 2021). Correlational research design is a non-experimental approach that a researcher can use to determine the interaction between variables (Sapp, 2017).

A causal-comparative research design could have been considered for this study if the intent was to compare the mean difference between groups. According to Brown (2020), causal-comparative research is conducted to allow a researcher to find a relationship after an event has already occurred and compares two or more groups. However, this research design was not

appropriate because the researcher was not concerned with comparing groups. A causalcomparative approach would be use for a study that utilizes ANOVA or a *t*-test analysis.

Definitions of Key Terms

Correlational research design. This research study utilized a correlational research design. The correlational research design implies that the research describes and measures the relationship between two or more variables in a study (Brown, 2020).

Heinrich's domino theory. Heinrich's domino theory is vital to this research study because it explains how safety can lead to job satisfaction. Heinrich's domino theory is a classical theory that suggests that injuries caused by an accident are the result of unsafe acts of a person or unsafe conditions at the workplace and are proceeded by the fault of a person and social environment (Osman et al., 2019).

Job satisfaction. Job satisfaction is the dependent variable for this study. Job satisfaction can be defined as how much an individual likes or dislikes their job (Spector, 1997). Another definition of job satisfaction is to what extent the employees are satisfied with their work (Eliyana & Ma'arif, 2019).

Occupational safety climate. Safety climate is used in this study with the added word "occupational" to represent the safety climate within an organizational environment. Safety climate can be defined as the way safety policies, procedures, and practices are perceived by an employee (Brown, 2020).

Transformational leadership. This study examined the relationship between transformational leadership, occupational safety climate, and job satisfaction. Transformational leaders can profoundly affect their followers and inspire followers to surpass their own interests for the organization's good (Gaviria-Rivera & Lopez-Zapata, 2019). In addition, transformational

leadership helps increase employee attention and strengthen the level of perception and acceptance of employees to the vision and goals of the group (Amalina et al., 2022).

Assumptions, Limitations, and Delimitations

Identifying the assumptions, limitations, and delimitations of a study allows the researcher to establish the scope and boundaries of the research. Assumptions are the assumed believes associated with this study. Limitations are unique to the methodology, design, and main areas of the research study parameters that are beyond the researcher's control (Roberts & Hyatt, 2018). Delimitations are the study's parameters. The following are the assumptions, limitations, and delimitations related to this research study.

Assumptions

According to Roberts and Hyatt (2018), an assumption is something a researcher takes for granted in their research. Assumptions are also expected to be accurate or at least probable by the researcher and the researcher's peers. A convenience sample from oil drilling and gas extraction industry workers located in the United States was taken for this study. The following assumptions are present in this study: All responses received are honest and express the responders' honest opinions, feelings, and values. It was also assumed that the sample is representative of the entire population that a linear relationship exists between the independent and dependent variables.

Limitations

Limitations are factors that affect the researcher's ability to generalize the results of a study and are also out of the researcher's control (Roberts & Hyatt, 2018). For example, a participant may not complete a survey in its entirety or answer truthfully. The first limitation is that the population was limited to oil drilling and gas extraction industry workers located in the

United States, which may affect generalizability. Also, representation of the entire population cannot be guaranteed because a convenience sample was used, The second limitation deals with the population also. Internet use increased from 52% in 2000 to 93% in 2021 (Wu et al., 2022). Even so, most oil drilling and gas extraction workers in the US work in the field rather than in an office setting where computers are readily available. Therefore, the sample's lack of technological access may have affected the survey return rate. The third limitation of this study is the length of the survey. The survey consisted of 131 items and Centiment.co estimated the time to complete the survey to be 20-30 minutes; however, some respondents may not have been able to complete the survey within that timeframe.

Delimitations

Delimitations clarify the study's boundaries and scope (Roberts & Hyatt, 2018; Theofanidis & Fountouki, 2018). The survey window was April 18, 2023, through May 3, 2023. Data collection was conducted online utilizing a convenience sample from the population of oil drilling and gas extraction industry workers in the United States. Another delimitation of this study was the sample size. The timeframe that which the survey was allowed to be open was a delimitation of this study. Additionally, the use of a quantitative correlational study design and method were delimitations of this study. Finally, the sample's perception of the study's variables may differ from what is identified in the study. For example, the NOSACQ-50 responses are based on the personal perspective, opinion, perception of the participant and therefore may not express these attributes of the everyone in the population. The study's variables are delimited to the dimensions discussed in the problem statement.

Summary and Organization of the Remainder of the Study

Chapter 1 included an introduction of this study on the relationship between transformational leadership, occupational safety climate, and job satisfaction. The background problem statement, purpose statement, research questions and hypothesis are also included in this chapter. The problem that was examined in this study is that many organizations ignore safety culture altogether, which can result in diminished job satisfaction (Niartiningsih et al., 2020). The purpose of this quantitative correlational study was to examine the relationship between transformational leadership, occupational safety climate, and job satisfaction of US oil drilling and gas extraction industry workers. Chapter 1 also contained information on advancing theoretical knowledge, the significance of the study, and the study's application to business management. This research is applicable to business because job satisfaction is directly related to turnover, motivation, and employee well-being. Leaders who are responsible for business practices and policies including leadership training, safety policies, and employee retention can use the results of this study to inform the implementation of strategies to enhance these factors. Chapter 1 included the study's methodology, research design, definition of terms, assumptions, limitations, and delimitations. Finally, for the purpose of advancing theory, the significance of this study lies in the contribution it will provide by expanding the literature and present knowledge about transformational leadership style and job satisfaction through occupational safety climate, a topic that requires further empirical investigation.

Chapter 2 includes a literature review on the variables of the study, theories, framework of the study, and the instruments used to complete the data collection for this study. Chapter 3 includes detailed information on the methodology and design of the study. In addition, the techniques used in the study that will allow future researchers to duplicate the process are

described. Chapter 4 includes the data collection and analysis of this study. The results of the data analysis are discussed. Finally, Chapter 5 includes a summary of the study as well as theoretical and practical implications and recommendations and concluding remarks.

CHAPTER 2: LITERATURE REVIEW

The aim of this study was to determine if there is a relationship between transformational leadership, occupational safety climate, and job satisfaction. It is possible to improve the effectiveness of management and organizational outcomes by gaining a better understanding of transformational leadership style and how it affects employee satisfaction and organizational performance. This chapter provides an overview of the theoretical foundations as well as the literature related to occupational safety climate, transformational leadership, and job satisfaction. Leadership research has been growing over the past decade, and a number of significant studies have been produced on the topic. However, due to leadership's broad and fragmented nature, leadership research remains primarily focused on the influence process between the leader and the employees (Lee et al., 2020). A literature review on transformational leadership, occupational safety climate, and job satisfaction is intended to provide a comprehensive overview of available information on transformational leadership, occupational safety climate and job satisfaction oil drilling and gas extraction industry workers.

Chapter 2 scrutinized extant literature and provided a background and overview of transformational leadership, occupational safety climate, and job satisfaction. To date, no studies or readily accessible information exists regarding the effects of transformational leadership on occupational safety climate and job satisfaction. A gap in the literature exists about how transformational leadership and occupational safety climate can enhance job satisfaction (Asgari et al., 2020). Other research shows that when the organizational culture is not aligned with the transformational leadership style, positive outcomes do not exist (Eliyana & Ma'arif, 2019).

There is a growing need for individual expertise across virtually all organizations; therefore, employees across all levels have become the most valuable assets within an

organization (Akdere & Egan, 2020). Now, because of this shift, human resources development focuses on providing managerial leadership training and support for employee learning and performance to achieve organizational mission success (Akdere & Egan, 2020). This research contributes to the literature on transformational leadership by further conceptualizing its effects on employee job satisfaction and an organization's occupational safety climate. Occupational safety climate perceptions and characteristics have also changed over time.

A review of the literature identified specific themes and topics related to transformational leadership, occupational safety climate, and job satisfaction. It was noted throughout the literature review that transformational leadership could significantly improve an organization's overall performance by improving employee commitment and behavior. These changes have the potential to affect the ability of a business to lower injury costs, improve job satisfaction, and increase employee retention rates and production (Shabane et al., 2017; Thompson et al., 2021; Waldman et al., 2012; Willis et al., 2017). These changes can also affect the amount of employee turnover and employee retention (Day et al., 2022).

A synthesis of the existing literature on these topics was generated from this historical review. Keywords used to identify relevant literature were transformational leadership, occupational safety climate, job satisfaction, leadership theory, safety theory, employee satisfaction, leadership roles and responsibilities, leadership questionnaire, job satisfaction survey, human resources management, safety climate scale, organizational commitment, management practice, safety climate, safety perceptions, theory-based safety climate instrument, correlation, corporate social responsibility, leadership analysis, psychosocial safety climate, workplace safety, leadership behaviors, and safety culture. The databases used to obtain

information include Business Source Ultimate, ProQuest, ScienceDirect, PLoS ONE, Academic Search Ultimate, JSTOR, and EBSCOhost.

Theoretical Foundations

The purpose of this section of the dissertation was to identify and explain the theories that form the basis of this research study. The theoretical framework of the present study was based on three main theories. These theories are Bernard Bass's transformational leadership theory, H. W. Heinrich's domino theory, and Abraham Maslow's hierarchy of needs theory. Understanding these theories is essential to understanding how a transformational leader is able to affect an employee's job satisfaction and an organization's safety climate. A better understanding of these theories requires an examination of the body of research that precedes this present study. Finally, the conceptual framework for this study and the theoretical foundation of the literature review starts with Bernard Bass's transformational leadership theory.

Bernard Bass's Transformational Leadership Theory

In recent years, an increasing amount of research has raised concerns about the validity of transformational leadership. In light of these concerns, further research is required in this area. Research regarding the topic of leadership has been a subject of theory for quite some time, but analytical research on leadership did not start until the beginning of the 20th century (AL-Nawafah et al., 2022).

Transformational leadership was first introduced by Downton in 1973 (Reza, 2019). In 1978, James MacGregor Burns, a well-known author, historian, and political scientist, proposed the idea of transformational leadership in his book *Leadership* (Barbinta et al., 2017; Reza, 2019). Finally, in his work on industrial psychology, Bernard Bass returned to the notion of transformational leadership in 1985 (Barbinta et al., 2017). Bernard Bass carried forward the ideas of James Burns and Abraham Maslow, which optimized the transformational leadership theory (Barbinta et al., 2017).

Bass's transformational leadership theory is characterized by a leadership style in which leaders clarify roles and task requirements for subordinates to achieve specific outcomes (Bass, 1985). As per the existing literature, transformational leadership is characterized by charismatic leadership traits that encourage followers to perform better than they would under other leadership styles (Day et al., 2022). In addition, a transformational leader's ability to influence subordinates is mediated by how performance change reinforcements are planned, when they are sent, and whether they are consistent or variable (Bass, 1985).

This study contributes to and adds content to the existing body of knowledge about the relationship between transformational leadership, organizational safety climate, and job satisfaction. Prior studies have not examined the effects of transformational leadership on occupational safety climate when examining the relationship between transformational leadership and job satisfaction. Furthermore, this study contributes to the existing body of knowledge on transformational leadership and adds contribution to the literature on transformational leadership's effect on job satisfaction and occupational safety climate. In addition, this study on transformational leadership adds content to the existing body of knowledge on the effects of transformational leadership adds content to the existing body of safety climate.

Research shows that transformational leadership results in increased attachment by subordinates to their organizations (Barbinta et al., 2017). Further investigation of the transformational leadership theory will help determine if the leadership style and occupational safety climate can help predict job satisfaction. It can be advantageous for business managers

and organizations to gain a better understanding of the transformational leadership style and how it impacts employee job satisfaction and organizational performance. Figure 1 is an illustration of the model of Bass's transformational leadership theory.

Figure 1



Bernard Bass's Transformational Leadership Model

Note. The transformational leadership model encourages, inspires, and motivates employees to innovate and create the changes necessary to shape the employee into a leader. From Transformational Leadership in Nursing," by Renjith et al. (2015).

Moreover, according to Bass's transformational leadership theory, leaders with transformational skills can provide both the economic and social benefits that employees expect, such as intellectual stimulation, individualized consideration to satisfy their needs, a model influence, and inspirational motivation that triggers intrinsic motivation in their employees (Wu et al., 2022). Finally, transformational leadership has long been observed to positively affect employees' behavior because it establishes trust and a greater sense of sensitivity toward employees' needs. Because the transformational leadership theory postulates that transformational leaders can motivate their subordinates and make them more satisfied with their jobs, the researcher conducted hypothesis testing to examine the relationship between transformational leadership, occupational safety climate and job satisfaction in the population of a sample of oil drilling and gas extraction industry workers in the United States. This study also required research on Heinrich's domino theory, which is discussed in the next section.

Heinrich's Domino Theory

A prominent accident prevention theory called the domino theory was established by H. W. Heinrich in the 1930s (Yorio & Moore, 2018). Heinrich utilized data collected from insurance claims to scientifically formulate the domino theory, which was published officially in 1931 (Dunlap et al., 2019). This foundational theory is the most influential accident prevention theory in the occupational safety and health discipline (Dunlap et al., 2019; Yorio & Moore, 2018). Results of studies conducted between 2003 and 2014 to determine the validity of the original theory show dynamic changes in the workplace; however, the basis of Heinrich's theory still remains (Bass et al., 2003; Dunlap et al., 2019). Furthermore, the findings of such research showed that while more information provides an in-depth view of accident causation, the original theory is in fact accurate and reliable (Dunlap et al., 2019).

Heinrich's domino theory places the focus on human error, claiming that most accidents are caused by people's unsafe actions (Akbari et al., 2019). A study based on Heinrich's theory suggests that human factors are the most important factors to investigate because they are responsible for most workplace accidents (Osman et al., 2019). Heinrich's domino theory also attributes 88% of accidents to a person's unsafe actions, 10% to unsafe conditions, and 2% to

unpredictable circumstances (Akbari et al., 2019). The focal point and overall conclusion to Heinrich's domino theory are that severe occupational safety and health incidents are preceded by many incidents or near misses of lower severity (Yorio & Moore, 2018).

Moreover, Heinrich's domino theory establishes that accident causation is based on the unsafe acts of individuals (Yorio & Moore, 2018). This study aimed to determine if the changes introduced by transformational leadership style in the workplace will result in a safer work environment and whether these variables together increase job satisfaction. The domino theory established the "safety triangle," as seen in Figure 2, which explains how serious incidents are proceeded by superficial incidents. Figure 2 also shows the indicators that can determine when a potential incident will occur. This study aimed to determine if there is a relationship between transformational leadership and occupational safety climate, which will establish the relationship to the safety triangle theory.

Figure 2

H. W. Heinrich's Safety Triangle



Finally, this study aimed to evaluate whether the relationship between transformational leadership and occupational safety climate results in a positive effect on employees' behavior that can result in greater job satisfaction based on a safer work environment. The results of this study contribute to the body of knowledge about Heinrich's domino theory and how it can be used to predict job satisfaction. This study also required further research on Maslow's hierarchy of needs theory, which is discussed in the next section.

Maslow's Hierarchy of Needs Theory

In 1943, Abraham Maslow's hierarchy of needs theory was seen in print form for the first time (McCleskey & Ruddell, 2020). Mr. Maslow continued researching and writing on the theory for ten years after the initial print appearance. The foundation of the hierarchy of needs theory is that there are five basic needs that are universal to all humans. Additionally, the theory implies that the needs at the current level must be met before the proceeding level can be satisfied. The hierarchy of needs theory can be found in a variety of scholarly media relating to topics ranging from leadership and management to psychology and behavior (McCleskey & Ruddell, 2020).

Maslow proposed that humans all have five universal needs that are arranged in order of importance, starting with physiological (Fallatah & Syed, 2018). The following are the remaining needs in order: safety, social, esteem, and self-actualization. These needs are common to people of all cultures (McCleskey & Ruddell, 2020). The hierarchy of needs establishes the basis of motivation for employees (Fallatah & Syed, 2018). Although Maslow's hierarchy of needs theory has been validated through multiple empirical studies, there are still some studies that bespeak criticism of the theory (Fallatah & Syed, 2018).

Figure 3

Abraham Maslow's Hierarchy of Needs Pyramid



Note. From "Maslow's Hierarchy of Needs," by Saul Mcleode (2022).

Review of the Literature

In the literature review, the theoretical framework of the study is presented, as well as the topic of the study. The main topics of this study were transformational leadership, occupational safety climate, and job satisfaction. The purpose of this literature review on transformational leadership, occupational safety climate, and job satisfaction was to provide a comprehensive collection of information regarding how transformational leadership influences occupational safety climate and job satisfaction. The results of previous research studies on transformational leadership have been inconsistent, resulting in a research gap (Kwang-rae et al., 2016; Ramakrishnan et al., 2020). Chapter 2 provides a thorough examination of the problem, along with an explanation of the theoretical framework used for this study and a description of the gap or need observed in the existing literature. This review examines the current state of the issues surrounding transformational leadership, occupational safety climate, and job satisfaction by reviewing the findings from recent research.

Transformational Leadership

The growing diversity of workforces and the desire to create an inclusive workforce have forced leadership approaches to change to manage the growing diversity levels within organizations (McCallaghan et al., 2019). Leaders are highly significant in terms of both their power and role within an organization, especially when it is hierarchical in nature (Amalina et al., 2022). Transformational leadership is no exception. In 1985, Bernard Bass developed the transformational leadership theory that stipulates transformational leaders influence subordinates based on the relationship and leadership qualities they encompass. By 2001, a constant concern regarding the new paradigm of transactional leadership had developed (Alimo-Metcalfe & Alban-Metcalfe, 2001; Barbinta et al., 2017). Even so, transformational leadership has been adopted by industries across the globe (Akdere & Egan, 2020).

Transformational leadership is demonstrated by leaders who motivate their followers beyond their individual self-interests for the benefit of the group, organization, and society (Muchiri et al., 2019). Transformational leadership theory states that a leader demonstrates the following characteristics: idealized influence, inspirational motivation, intellectual stimulation, and individualized consideration (Muchiri et al., 2019). A leader who practices idealized influence leadership behaves in a way that will make them a role model for their followers, putting their followers' needs above their own, sharing the risks with their followers, and taking their followers' needs into account (Muchiri et al., 2019). A leader with idealized influence also displays high standards of moral conduct in his or her own conduct (Inness et al., 2010). There are many different types of motivation, but inspirational motivation occurs when leaders communicate to employees a positive, value-based vision of the organization's future state and how it can achieve that goal. By providing meaning and challenge to their followers in their

work, inspirational leaders can motivate and inspire their followers through their leadership style (Muchiri et al., 2019). An example of intellectual stimulation by a transformational leader is when an employee challenges organizational norms and is encouraged to think creatively (Inness et al., 2010). Inspirational leaders can stimulate intellectual curiosity, which leads leaders to impart a sense of innovation and creativity in their followers (Muchiri et al., 2019). inspires them to question assumptions, reframe problems, and approach old situations in new ways through intellectual stimulation (Muchiri et al., 2019). The final characteristic that emulates transformational leadership behavior is the concept of individual consideration. Individual consideration is one of the ways in which leaders recognize their followers' unique needs (Inness et al., 2010). Finally, leaders who are individually considerate pay special attention to the individual's growth and achievement needs by acting as a coach or mentor for the individual, ensuring that his or her needs are met (Muchiri et al., 2019).

An important aspect of transformational leadership is the ability of leaders to motivate their employees and inspire them to perform at their best by utilizing a variety of methods (Akdere & Egan, 2020). Human resources development is an example of a method deployed by organizations to teach transformational leadership skills to their employees. When appropriately used, transformational leadership skills, in conjunction with other human resources development training and education lead employees to successfully develop common goals (Akdere & Egan, 2020).

Transformational leaders also promote moral development in employees (Barbinta et al., 2017). Additionally, transformational leaders encourage employee dedication to the organization and commitment to leaders (Akdere & Egan, 2020). This dedication and commitment promote the overall success of the organization's goals (Akdere & Egan, 2020). Finally, leaders who
promote high levels of safety climate provide a signal to employees about prioritization of safety over production goals which can help the organization's overall culture (Lyubykh et al., 2022). This element of leadership is just one way that transformational leaders can have an impact directly on the employees and the organizational culture.

Researchers have long recognized the influence of psychosocial work factors on organizational behavior (Gyekye & Haybatollahi, 2014). Dollard et al. (2019) conducted research on the psychosocial safety climate and authored a book describing the importance of the leading indicators of the psychosocial safety climate for employees. Psychosocial work factors include the climate or culture in the workplace, work roles, and task design and content (Gyekye & Haybatollahi, 2014). As a result of promoting psychological health over productivity, organizations with a high psychosocial safety climate improve the work environment by reducing demands, increasing resources, providing job security, adequate salaries, and instituting policies against bullying and violence (Loh et al., 2021). Reduced stress promotes employee psychological health, which in turn, boosts performance. Transformational leaders can also help reduce stress in their followers and improve leadership quality and reduce risks in the workplace (Berthelsen et al., 2020). Poor leadership and elevated risks are known as the "cause of the causes," and can lead to unsafe working conditions. Reduced stress promotes employee psychological health, which in turn boosts performance.

Organizations spend billions of dollars each year on leadership training (Badura et al., 2021). This expense is one example of how vital leadership training is for organizations. Transformational leadership also affects employees' perception of the importance of organizational programs, such as the organization's safety climate (Wu et al., 2022).

Transformational Leadership's Effects on Occupational Safety Climate

Often, leaders are seen as crucial players in the creation of localized climates within various levels of an organization by acting as "interpretive filters" (Flatau-Harrison et al., 2020). Employees' safety perceptions might be influenced by the safety climate and the organization's safety management practices (Osman et al., 2019). Consequently, transformational leadership has been related to safety behavior, safety culture, and safety climate (Muchiri et al., 2019).

The transformational leader allows subordinates to think creatively and allows employees to feel empowered by allowing their voices to be heard (Day et al., 2022). This empowerment allows employees to act as change agents for the organization (Day et al., 2022). Such change can promote positive change to the occupational safety climate by allowing for a proactive approach to safety rather than a reactive approach (Day et al., 2022). Research by Wu et al. (2022) yielded equivocal results that indicate transformational leadership has a positive relationship with occupational safety compliance. The research results also indicated that employees possess an obligation toward their transformational leaders.

On the other hand, Wu et al. (2022) also determined that transformational leadership can negatively impact safety compliance via a tolerance for unnecessary risks. Contrarily, research by Flatau-Harrison et al. (2020) resulted in findings that show that within safety-specific contexts, there has been strong support for the role of transformational leadership in promoting a positive safety climate within the workplace. Transformational leaders' positive safety climate promotional behavior also promotes safety participation and increased safety behavior in employees (Flatau-Harrison et al., 2020). It is essential for leaders to be motivated to implement safety rules and policies (Kaluza et al., 2020). In the research on safety climate, it has been demonstrated that the safety climate of an organization is significantly influenced by the level of

support for safety by its leaders (Kaluza et al., 2020). Transformational leaders have the motivational characteristics required for this type of implementation and to make the implementation successful.

Research by Dunlap et al. (2019) shows that factors within the organization can lead to significant changes in the organizational climate. Transformational leadership can be an instigator of such changes. An example would be a relaxed safety climate or an elevated risk tolerance, resulting in a hindered occupational safety climate. (Wu et al., 2022). On the other hand, transformational leadership can result in an enhanced occupational safety climate through sustained organizational commitment in workers. When employees perceive safety as an organizational priority, they are more likely to possess safety knowledge, be willing to engage in safety behaviors (safety motivation), and enact higher levels of safety behavior, thus reducing the risk tolerance necessary in the organization. The importance of leadership in occupational safety cannot be overstated. There is a growing recognition that transformational leadership is a central component of leadership that promotes safety (Grill et al., 2019). In the transformational leadership theory, individuals are considered, intellectual stimulation is provided, inspirational motivation is evoked, and idealized influence is used to enhance the culture of a company (Akdere & Egan, 2020).

Additional factors can play a significant role in a transformational leader's ability to be effective (Willis et al., 2017). Employees' perceptions of the risk of an accident within their work context are one of these factors. Therefore, risk perception can affect a transformational leader's effectiveness (Willis et al., 2017). Overall, this research contributes to business management by showing that transformational leadership can be both positive and negative and that leaders need to ensure that the proper safety perception is instilled in the work environment.

Occupational Safety Climate

Taking care of the health and safety of its employees should be a top priority for any company. In fact, since general climate measures are considered too broad and unfocused to predict specific outcomes, interest in specific aspects of organizational climate has grown in recent years (Kaluza et al., 2020). To better understand occupational safety climate, this study explains the concepts that make up the occupational safety climate. *Safety* is a broad term that refers to a legal order that society can exist under without fear (Calis & Buyukakinci, 2019). Occupational safety is a relatively new term that was introduced in 2019 in conjunction with industrialization. *Safety climate* can be defined as workers' perceptions of policies, procedures, and practices related to safety in any given organization (Wu et al., 2022). Furthermore, an employee's occupational safety climate is concerned with protecting an employee while at work against the dangers associated with the job they do for a living (Calis & Buyukakinci, 2019).

Several studies have shown that the safety climate in an organization can influence accidents as a leading indicator or as a lagging indicator (Willis et al., 2017). In terms of workplace safety, leadership's perceived commitment to safety is a particularly meaningful indicator (Beus et al., 2017). Safety climate perceptions and subsequent group safety climates are often influenced by the actions of leaders that reflect their commitment to safety (or lack thereof). Because of its influence on safety, leadership commitment and the indicators that follow result in employee knowledge and motivation (Flatau-Harrison et al., 2020). Thus, the organizational safety climate is associated with self-reported safety compliance and participation behaviors (Flatau-Harrison et al., 2020).

Despite the increase in safety climate research in recent years, conceptual ambiguity persists, which both raises questions about what safety climate actually is and interferes with a

deeper scientific understanding of the concept (Beus et al., 2017). Since the early 1990s, occupational risk prevention has been the driving force behind workplace safety (Sánchez-Segura et al., 2021). In contrast, the global COVID-19 pandemic resulted in a new standard of normal work environment, and workers often work digitally rather than in person. In occupational safety literature, the causes of occupational accidents have been classified as unsafe working conditions or unsafe workers' behavior, regardless of location (Akbari et al., 2019). Working conditions are one component of the occupational safety climate.

The transformational leadership theory specifies that team members change their attitudes, behaviors, and beliefs while adhering to higher goals when a deep connection between the leader and his team is in place (Barbinta et al., 2017). This change in attitude and behavior can result in a safer workplace and reduce the number of workplace injuries. Nevertheless, there have been no studies investigating and dissecting how transformational leadership affects occupational safety, despite the range of outcomes it has (Barling et al., 2002). The third and final component of this study is job satisfaction. Researchers recommend that future research be conducted to assess if changes in transformational leadership lead to changes in employees' occupational safety (Barling et al., 2002). The significance of this study lies in the contribution it will make to expand the literature and present knowledge about transformational leadership style and job satisfaction through occupational safety climate, a topic that requires further empirical investigation.

Perceived Organizational Support Effects on Occupational Safety Climate

According to Tan et al. (2021), there is a relationship between an individual (the worker) and an organization (the workplace). Perceived organizational support encourages employees to have a greater work commitment, which promotes in-role behavior that is conducive to the

successful operation of the organization (Tan et al., 2021). Workplace safety perceptions and other organizational factors that affect workplace safety appear to be closely related to perceived support levels in an organization (Gyekye & Saiminen, 2007). It is not entirely guaranteed that a safe work environment will be established by having an occupational health and safety management system (Calis & Buyukakinci, 2019).

Safety climate is considered a subset of organizational climate. Moreover, safety climate also refers to the reasonable set of perceptions and expectations that workers have regarding safety in their organization (Gyekye & Saiminen, 2007). This climate is unstable and subject to change as conditions shift in an organization (Hall et al., 2013).

A key aspect of organizational behavior that impacts how workers perceive safety at work and, in turn, how they behave at work is the extent to which they perceive their organization as supportive and caring (Gyekye & Saiminen, 2007). Transformational leaders display this type of supportive and caring behavior which has been connected to a positive relationship to organizational safety climate. Employees also need to see that the organization's upper management expects adherence to safety policies (Hall et al., 2013). Manager/supervisor support, such as transformational leadership behavior, reinforces the concept of a top-down approach to positively influencing safety climate (Hall et al., 2013).

Corporate social responsibility activities show the public a company's interest in a multitude of different issues. Occupational safety climate is just one of these issues on which corporate social responsibility focuses. In terms of work safety, corporate social responsibility research indicates that these activities enhance the development and support of health and safety practices within an organization (Kubasinski & Slawinska, 2021). Therefore, improving the workplace safety climate is a critical component of corporate social responsibility activities, also

known as proactive activities (Kubasinski & Slawinska, 2021). It is also widely accepted that proactive activities are directly related to transformational leadership characteristics and can lead to a more proactive safety climate rather than a reactive safety climate, leading to higher job satisfaction.

Job Satisfaction

Job satisfaction is a very complex situation in which the source of satisfaction can come from multiple directions, such as salary, promotion opportunities, supervision, colleagues, and many others (Jauhar et al., 2017). In conjunction with the satisfaction source, many other factors also apply to job satisfaction (Niartiningsih et al., 2020). For example, there are significant differences in generational preferences. Generation Y employees have different views on job longevity, which affect their opinions on job satisfaction (Jauhar et al., 2017).

There are also many ways to define a person's job satisfaction. For instance, Akbari et al. (2019) define job satisfaction as people's positive attitude toward their job. In the context of employment, job satisfaction can also be defined as a state of pleasure or happiness that arises from being satisfied with one's current employment situation and includes all aspects of that employment situation. In this study, to adequately evaluate an employee's job satisfaction, both positive and negative aspects must be considered (Niartiningsih et al., 2020).

The concept of job satisfaction can be divided into two dimensions: intrinsic satisfaction and extrinsic satisfaction (McCallaghan et al., 2019). In addition to autonomy, variety of skills, supervision, and degree of responsibility, intrinsic satisfaction is related to qualitative or subjective elements of a job (McCallaghan et al., 2019). For example, employees are more likely to experience intrinsic satisfaction when they receive incentives, rewards, promotions, and safe working conditions (McCallaghan et al., 2019).

In the workplace, an individual is satisfied because they are motivated by factors and conditions (Eliyana & Ma'arif, 2019). The simplest way to express job satisfaction is to say it is an emotional sensation resulting from the work done by an individual. A person's job satisfaction can also depend on how satisfied he or she is with the job duties and working conditions (Akbari et al., 2019). Moreover, one of the most important factors influencing the productivity and efficiency of human resources is the degree to which job satisfaction correlates with job responsibilities, accountability, actions, responses, motivations, and encouragement.

Borell et al. (2019) conducted research on job satisfaction which shows that many authors had shared views on job satisfaction (). The outcome of initial research led researchers to believe that job satisfaction was explained by contextual factors such as relationships with leaders and organizational culture, among others. Additionally, leaders can promote healthy job satisfaction levels by providing signals to the employees that safety has priority over production goals (Lyubykh et al., 2022). According to research, a positive correlation exists between a high level of job satisfaction among employees and their commitment to a workplace environment where occupational learning is emphasized (Lin et al., 2021). Moreover, researchers recommend future research on the effect of leadership profiles on job satisfaction (Borell et al., 2019).

Research by Guarnaccia et al. (2016) found that occupational self-efficacy can significantly affect a worker's job satisfaction. Self-efficacy can be defined as an individual's beliefs about their capabilities to produce designated levels of performance (Guarnaccia et al., 2016). Streamlined research into occupational self-efficacy, which is self-efficacy in the workplace, found that job satisfaction is also affected by organizational commitment factors such as management and performance feedback (Guarnaccia et al., 2016). As an indication of what managers should do in practice, other individual variables, such as determinants of job

satisfaction, should be considered (Borell et al., 2019). When human resources departments connect managerial leadership support with employee learning and performance, employee job satisfaction levels increase (Akdere & Egan, 2020).

Job satisfaction can also depend on the employee's ability to achieve certain goals (Steinmann et al., 2018). Transformational leaders directly influence an employee's perception of the achievability of such goals (Steinmann et al., 2018). This influence allows affected employees to achieve job satisfaction. Nevertheless, there are yet to be readily available studies to show the effect of transformational leadership and occupational safety climate on job satisfaction. Research by Eliyana and Ma'arif (2019) found that there is a significant direct effect on job satisfaction when transformational leadership styles are present in the workplace. However, their research study also concluded that when confronted with organizational commitment, transformational leadership cannot directly impact work performance when the two factors are in discordance (Eliyana & Ma'arif, 2019). Further research is needed to examine how the occupational safety climate and leadership affect employee job satisfaction.

There is research on job satisfaction regarding Maslow's hierarchy of needs. According to Maslow's hierarchy of needs theory, employees need safety, love, support, and confidence (Fallatah & Syed, 2018). The employing company is vital to this process, as are the leaders within the company. Again, generational differences can affect what the employee needs to be satisfied (Jauhar et al., 2017). Employees who achieve these needs are more likely to achieve job satisfaction than those employees whose needs are not met.

Maslow's hierarchy of needs also stipulates that employers should recognize employees' esteem needs (Fallatah & Syed, 2018). The concept of self-esteem refers to the requirement to respect one's rights, acknowledge one's accomplishments, and recognize one's autonomy and

independence. Without these esteem needs being met by the employer, employees may never reach job satisfaction (Fallatah & Syed, 2018).

Turnover Intention and Job Satisfaction

Turnover is costly to organizations and is one of the main reasons managers and business leaders focus on job satisfaction in their organizations (Ozkan et al., 2020). Job satisfaction is chosen as one of the most important predictors of turnover intention, followed by organizational commitment and self-actualization. (Ozkan et al., 2020). It is generally believed that employee satisfaction is influenced by how productive, creative, and committed they are at their workplace. It is also thought that it is a major factor influencing their intention to quit. (Jauhar et al., 2017). Because of this, job satisfaction has been linked to high turnover rates for organizations (Lin et al., 2021). On the contrary, higher levels of job satisfaction leads to a reduction in turnover intention (Jauhar et al., 2017; Lin et al., 2021).

Job satisfaction is one of the most critical factors that impact the effectiveness of an organization. Job satisfaction is susceptible to being influenced by the corporate culture and leadership of the organization (Jauhar et al., 2017). Workers who are satisfied with their job or have higher levels of job satisfaction have been shown to have higher levels of organizational commitment (Lin et al., 2021). Higher levels of job satisfaction can, in turn, reduce the amount of employee turnover and thus reduce costs to the employer (Lin et al., 2021). Research conducted by Amalina et al. (2022) aimed to discover how transformational leadership affects employee performance mediated by job satisfaction. The research by Amalina et al. revealed a positive and significant influence between transformational leadership on job satisfaction. Another study found that the absence of a transformational leader's promotion of new ideas in the workplace also led to decreased job satisfaction among employees (Kasztelnik & Al-ahmadi,

2022). Researchers also found a positive and significant influence of transformational leadership on employee performance through job satisfaction.

Methodology

There is a great deal of literature examining transformational leadership, workplace safety climate, and job satisfaction. This large body of literature examines transformational leadership, occupational safety climate, and job satisfaction in detail. Yet none of the studies researched tested the relationship between transformational leadership, occupational safety climate, and job satisfaction. Several examples of these studies are included next.

In 2020 a quantitative study was conducted by Akdere & Egan that examined safetyspecific transformational leadership and occupational safety climate. Additionally, Lyubykh et al., (2022) completed a study to determine the importance between leadership and safety climate. In 2022, a study by Summers et al., was conducted utilizing the NOSACQ-50 to evaluate safety climate. Another study by Batura et al., (2016) examined the job satisfaction survey. It is, however, rare for studies to investigate if there is any correlation between transformational leadership and occupational safety climate for oil drilling and gas extraction industry workers in the United States.

Instrumentation

Transformational leadership was assessed using the Multifactor Leadership Questionnaire (MLQ 5x-short). The MLQ 5x-short evaluates the full range of transformational leadership (Alimo-Metcalfe & Alban-Metcalfe, 2001). This full range includes nine different aspects of a leader (Alimo-Metcalfe & Alban-Metcalfe, 2001). The four main transformational components are idealized influence, inspirational motivation, intellectual stimulation, and individualized consideration. Idealized influence and inspirational motivation were later combined to create the

dimension of charismatic inspiration. The MLQ 5x-short utilizes a 45-question questionnaire (Alimo-Metcalfe & Alban-Metcalfe, 2001). In addition, the MLQ 5x-short utilizes a 5-point behavior scale to measure the participants' responses. Wu et al. (2022) found a relationship between leaders and safety utilizing the MLQ 5x- short survey.

Occupational safety climate was assessed utilizing the Nordic Occupational Safety Climate Questionnaire (NOSACQ-50). Similar researchers conducted tests to evaluate the NOSACQ-50 for reliability. The Cronbach alpha for a study conducted by Summers et al. (2022) was found to be good for the NOSACQ-50 dimensions. The NOSACQ-50 questionnaire consists of 50 items across seven safety climate dimensions and measures participants' shared perceptions of management safety, management safety empowerment, management safety justice, employees' safety commitment, employees' safety priority, employees' safety risk nonacceptance, safety communication, safety learning, trust in co-workers' safety competence, and trust in the efficacy of safety systems (Summers et al., 2022). The researcher was granted permission to use a verified and reliability-tested shortened version of the Nordic questionnaire to keep the survey for this study manageable.

The job satisfaction variable in this study was assessed using the Paul Spector Job Satisfaction Survey, which demonstrates beneficial properties for job satisfaction research studies (Lin et al., 2021). The Paul Spector Job Satisfaction Survey utilizes Paul Spector's Job Satisfaction Survey is made up of 36-questions. The survey utilizes a six-point scale that measures job satisfaction in the following ways. Each item is scored from one to six if the original response choices are used. High scores on the scale represent job satisfaction for the respondent. Low scores on the scale represent a lack of job satisfaction. Paul Spector Job Satisfaction Survey, utilizes questions to determine the person's attitude toward their job and

different aspects that are related to job satisfaction (Lin et al., 2021). This survey has been found to assess job satisfaction effectively and will sufficiently provide job satisfaction results for the present study (Bowling & Zelazny, 2021). In 2016, a similar study was conducted by Batura et al., that found that the JSS was a reliable, valid, and internally consistent.

Summary

Chapter 2 included information retrieved from multiple scholarly literature documents on the importance of, history of, and background of transformational leadership in organizations. Further information on the history, background, and importance of occupational safety climate and job satisfaction is also referenced. The information presented in this chapter utilized multiple databases and online sources of scholarly information regarding the primary variables of the study. Chapter 2 also included information on the theoretical framework for this study. Finally, this chapter included background information on transformational leadership, occupational safety climate, and job satisfaction that further educates readers on the history of these variables.

CHAPTER 3: METHODOLOGY

Two major concerns for business leaders are job satisfaction and the general safety of their workers. The purpose of this quantitative correlational study was to examine the relationship between transformational leadership, occupational safety climate, and job satisfaction. Participants for this study were recruited by Centiment from within the oil drilling and gas extraction industry in the United States.

Chapter 3 includes a review of methodology and a discussion of the methods of selecting instruments and procedures for collecting and analyzing data. Measures regarding the validity and reliability of the research instruments are discussed. A discussion of ethical considerations pertaining to this research study will bring this chapter to a close.

Statement of the Problem

An employee's safety while at work is the responsibility of the employer (OSHA, 1970). Different leadership styles can affect the occupational safety climate of an organization. In turn, this can affect a worker's job satisfaction. The problem that will be examined in this quantitative study is that many organizations ignore the safety climate altogether, which can result in diminished job satisfaction (Niartiningsih et al., 2020). This study highlighted the impact leadership and occupational safety climate have on employee job satisfaction. The findings from this study may encourage leaders to consider the impact of these variables on employees in their organization. Leaders can use the results of this study to create and implement strategies to increase production, reduce injuries, create a safer work environment, and increase job satisfaction.

Research Questions and Hypotheses

The purpose of this quantitative correlational study was to examine the relationship between transformational leadership, occupational safety climate, and job satisfaction. There is one research question and three sets of hypotheses to address the study's purpose.

RQ: What is the relationship between transformational leadership, occupational safety climate, and job satisfaction of gas and oil industry workers in the United States?

H1₀: There is no significant relationship between transformational leadership and job satisfaction of oil drilling and gas extraction industry workers in the United States.

H1_a: There is a statistically significant relationship between transformational leadership and job satisfaction of oil drilling and gas extraction industry workers in the United States.

H2₀: There is no significant relationship between perceived occupational safety climate and job satisfaction of oil drilling and gas extraction industry workers in the United States.

 $H2_a$: There is a statistically significant relationship between perceived occupational safety climate and job satisfaction of oil drilling and gas extraction industry workers in the United States.

H3₀: There is no statistically significant relationship between transformational leadership and perceived occupational safety climate of oil drilling and gas extraction industry workers in the United States.

H3_a: There is a statistically significant relationship between transformational leadership and perceived occupational safety climate of oil drilling and gas extraction industry workers in the United States.

Research Methodology

As part of the scientific method, quantitative methods are used to examine the relationships between variables without making any inference to a causal relationship between them (Le Tourneau University, 2022). A quantitative, non-experimental correlational design was used in this study and hypothesis testing was conducted to examine the relationship between transformational leadership, occupational safety climate, and job satisfaction.

This study could potentially be conducted using qualitative methods; however, a qualitative method was not chosen primarily because hypotheses testing using validated and reliability-tested psychometric survey instruments was conducted that allowed the researcher to numerically examine relationships between the variables being investigated. The purpose of this research was not to explore people's perceptions, thoughts, and feelings and analyze emerging themes. In addition, qualitative approaches are often regarded as theoretical in nature rather than practical or applied, as is quantitative research. Therefore, a qualitative approach was not appropriate for this study.

The researcher recruited 107 members of the target population through convenience sampling to participate in the study. Convenience sampling is often used in quantitative research studies because it targets members of a population who are easily accessible to the researcher and are willing to participate in a research study (Brown, 2020). It is a method of nonprobability or nonrandom sampling. The population for this research study was workers in the oil drilling and gas extraction industry located in the United States. The survey was open from April 18, 2023, through May 3, 2023.

Research Design

This study utilized correlational design. Correlational research is used to examine relationships between variables that can be quantified numerically (Limberg et al., 2021; Ojeda, 2023). Correlational design was chosen specifically for its ability to investigate whether there is a linear relationship between the variables by comparing two sets of data (CITI Program, n.d.). Correlational research is a popular form of research because it makes it possible for the researcher to examine constructs that are not directly observed by the researcher (Limberg et al., 2021). As part of the evaluation process, transformational leadership was evaluated using the MLQ 5x-short, occupational safety climate was evaluated using the NOSACQ-50, and job satisfaction was assessed using the JSS.

Population and Sample Selection

The target population for this study was oil drilling and gas extraction industry workers in the United States. To be eligible to participate in this study, individuals must work in the industry and be at least 18 years old. The link to a psychometric survey instrument was provided to participants and data was collected through deliberate sampling. Centiment was contracted to assist with online data collection, which allowed participants from various locations to participate in the study. The sample for this study was a convenience sample. The sample consisted of 107 participants. A G*Power analysis indicated that the minimum number of usable responses necessary to avoid committing a type II error was 102 (see Figure 4 below). The Bonferroni correction was applied to statistical testing and therefore a p-value of .025 was used to determine statistical significance due to the use of the variables twice in the correlation.

Figure 4

Sample Size Determination using G*Power



Centiment identified, contacted, and recruited participants who work in the oil drilling and gas extraction industry in the United States to complete the survey. Those who were not employed in the industry and those who are under 18 years of age were not permitted to participate.

Instrumentation

This study used three instruments consolidated into one survey to determine the relationship between transformational leadership, occupational safety climate, and job

satisfaction on a convenience sample of US workers in the oil drilling and gas extraction industry. The survey was expected to take approximately 20-30 minutes to complete.

Using the instruments that measure the constructs researchers are interested in studying allows them to get valid results. Transformational leadership was assessed using the MLQ 5x-short survey, occupational safety climate was assessed using the NOSCAQ-50 questionnaire, and job satisfaction was assessed utilizing the JSS survey. Informed consent instructions were provided for the respondents as part of the introduction to the online survey. Participants were required to consent before they were permitted to complete the survey.

The MLQ 5x-short originated in 1985 (Alimo-Metcalfe & Alban-Metcalfe, 2001). It is composed of 45 questions to assess participants' specific leadership behaviors and will be included in Appendix B (Statistic Solutions, n.d.). Academic research has recognized the validity of the MLQ 5x-short questionnaire (Wu et al., 2022). The MLQ 5x-short utilizes a five-point behavioral scale that allows responses that range from "Not at all" to "Frequently if not always" (Statistic Solutions, n.d.). Permission to use the MLQ 5x-short was obtained through Mindgarden.com. The MLQ 5x-short creators, Bernard Bass and Bruce Avolio, have given Mindgarden.com special permission to reproduce the MLQ 5x-short. At Mindgarden.com, researchers have the option to purchase a remote online survey license to distribute the survey via a non-Mind Garden survey system. This permission includes the right and license to use and administer the MLQ 5x-short online.

The NOSCAQ-50 questionnaire is an academically accepted, valid, and reliable survey to measure occupational safety climate (Grill et al., 2019; Kines et al., 2011). The NOSCAQ-50 is composed of 50 questions that measure seven safety climate dimensions. This questionnaire is designed to diagnose occupational safety climate and evaluate safety climate interventions

(NFA.dk - Det Nationale Forsknings Center for Arbejdsmiljø, n.d.). The researcher has obtained permission to download and utilize the copyrighted NOSACQ-50 questionnaire.

The JSS was developed in 1985 by Paul Spector. The JSS is a 36-item, nine-facet scale to assess respondents' attitudes about their job (Spector, 1997). Employee job satisfaction has been measured by the JSS in the USA, Singapore, Turkey, Pakistan, Taiwan, and Iran (Batura et al., 2016). This survey utilizes four items to assess each of the nine facets. Six choices are available per item, ranging from "strongly disagree" to "strongly agree." The composite survey instrument, including demographic questions, can be found in Appendix B.

Validity and Reliability

Research instruments are often described as being reliable and valid according to the psychometric characteristics a research instrument possesses (Andrade, 2018). In research, validity also applies to the study's findings. There are two types of validity, internal and external validity, which determine whether an instrument performs as expected (Andrade, 2018). Furthermore, four types of validity can be associated with correlation research studies. They are convergent, criterion, predictive, and face validity (Rapeli, 2022).

The concept of credibility in research is known as internal validity. Credibility refers to the fact that the study design, the study conduct, and the analysis of the findings answer the research questions without bias (Andrade, 2018). External validity is also described as transferability. In terms of external validity, the study findings are examined against the possibility of generalizing them to other situations.

Convergent validity is defined as the extent to which survey proxies measure the same construct (Rapeli, 2022). Next is criterion validity, which measures the extent to which the various proxies are associated with other variables that researchers would expect proxies to be

associated with based on knowledge of the survey subject. A survey's predictive validity refers to the extent to which the proxies are accurately able to predict the knowledge of the survey subject matter of the respondents. Lastly, face validity refers to whether a variable is presented on the surface as having enough credibility to measure what is supposed to be measured adequately The MLQ 5x-short was developed by Bruce J. Avolio and Bernard M. Bass (Alimo-Metcalfe & Alban-Metcalfe, 2001). While leadership scholars have focused a great deal of attention on the MLQ 5x- short, little interest has been shown in exploring the relationships between its underlying constructs and measures (Batista-Fogeut et al., 2021). According to Batista-Fogeut et al., the MLQ is the leading survey instrument used to examine leadership theory. Several metaanalyses have confirmed the MLQ's effectiveness and validity, making the instrument a prominent example of academic and commercial success. Table 1 shows correlations among the five MLQ 5x-short subdimensions scores from the 2004 study conducted by Avolio and Bass (2004). The study's results demonstrate a clear construct validity. The average reliability score for the transformational leadership characteristics was .76, which is higher than the reliability value for excellent. In addition, the average results of the study demonstrate clear construct validity, with the average intercorrelation scores among the five transformational scales being .64, thus further verifying the validity of this test.

Table 1

	II(A)	II(B)	IM	IS	IC	CR	MBEA	MBEP	LF	EE	EFF	SAT
II(A)	(.75)											
II(B)	.64**	(.70)										
IM	.68**	.68**	(.83)									
IS	.64**	.59**	.59**	(.75)								
IC	.71**	.60**	.59**	.68**	(.77)							
CR	.67**	.61**	.62**	.61**	.68**	(.69)						
MBEA	07**	.02**	.08**	01	12**	.01	(.75)					
MBEP	36**	27**	30**	33**	32**	32**	.10**	(.70)				
LF	49**	34**	37**	39**	42**	44**	.08**	.61**	(.71)			
EE	.71**	.57**	.62**	.62**	.68**	.63**	06**	33**	42**	(.83)		
EFF	.73**	.56**	.60**	.63**	.67**	.67**	06**	43**	56**	.72**	(.82)	
SAT	.75**	.54**	.60**	.62**	.70**	.64**	12**	40**	52**	.71**	.79**	(.79)

Intercorrelations among MLQ 5x-short Factor Scores

The NOSACQ-50 was used to assess the occupational safety climate for this study. Several instruments are available to measure safety climate; however, very few have proven capable of presenting a consistent factor structure across different contexts like the NOSACQ-50 (Kines et al., 2011). An international study was used to acquire empirical results and to test the validity of the questionnaire. This study was based on organizational and safety climate theory, psychological theory, previous empirical research, and a continuous learning process (Kines et al., 2011).

Table 2 shows the content and reliability results of the seven NOSACQ-50 safety climate scales. To estimate internal consistency reliability, a single measurement tool is usually administered to a group of people at one time (Abbas & Khanam, 2020). Cronbach's alpha is the best and most popular reliability index. The Cronbach's alpha scores of the NOSACQ-50 seen in Table 2 indicate validity and reliability. Based on the internal consistency in each subsample results, NOSACQ-50 was found to measure safety climate and predict safety motivation, perceived safety level, and self-rated safety behavior accurately (Kines et al., 2011). Furthermore, the NOSACQ-50 validity is confirmed based on its ability to detect significant differences in safety climate between organizational units.

Table 2

Content and Reliability of the Seven Safety Climate Scales

Scale	Number	Internal consistency	Internal consistency in each subsample (alpha)				
	of items	(Cronbach's alpha)	DK	FI	IC	NO	SE
1. Management safety priority, commitment and competence	9	0.87	0.84	0.89	0.89	0.85	0.88
2. Management safety empowerment	4	0.73	0.74	0.79	0.81	0.63	0.67
3. Management safety justice	4	0.71	0.60	0.79	0.74	0.68	0.72
4. Workers' safety commitment	6	0.77	0.80	0.84	0.79	0.71	0.73
5. Workers' safety priority and risk non-acceptance	7	0.80	0.77	0.81	0.81	0.82	0.77
6. Safety communication, learning, and trust in co-worker safety competence	8	0.79	0.76	0.75	0.85	0.80	0.76
7. Workers' trust in the efficacy of safety systems	7	0.82	0.80	0.85	0.81	0.79	0.83

Finally, the JSS, which was developed by Paul Spector in 1985, was utilized to assess job satisfaction. The JSS is a composite scale that has been peer reviewed and has been tested for reliability multiple times to confirm its validity (Abbas & Khanam, 2020; Spector, 1985). Table 3 shows the results of internal consistency and reliability of the JSS as conducted by Spector (1985), which indicate that the scale is reliable and internal consistency is good.

Table 3

Internal Consistency, Split-Half and Test-Retest Reliability of JSS

Subscale	Mean	SD	Mean interitem correlation	Coefficient alpha	Test-retest reliability
Pay	10.5	5.1	43	75	45
Promotion	11.5	5.1	40	73	62
Supervision	19.9	4.6	53	82	55
Benefits	13.1	5.0	40	73	37
Contingent rewards	13.4	5.1	44	76	59
Operating procedures	12.5	4.6	29	62	74
Co-workers	18.8	3.7	33	60	64
Nature of work	19.2	4.4	50	78	54
Communication	14.0	5.0	38	71	65
Total satisfaction	133.1	27.9	21	91	71
n	3,067	3,067	2,870	2,870	43

Table III. Means, Standard Deviations, and Reliabilities for the JSS

Data Collection Procedures

A total of three data collection instruments were combined into a single survey instrument and distributed by Centiment to collect data. To assess the level of transformational leadership in an organization, the MLQ 5x-short instrument was used, the NOSCAQ-50 was used to evaluate an organization's safety climate, and the JSS was used to measure job satisfaction.

The Centiment team facilitated the distribution of the survey instrument according to the inclusion criteria set for the target population. As part of its outreach to recruit participants during the open window for data collection, Centiment notified the target population via batches, soliciting from its prebuilt audience panel databases. To confidently conduct the statistical analysis, 107 survey responses were collected from the target population, which is higher than the sample size calculation referred to in the G*Power calculation found in Figure 4 above. The participants who are oil drilling and gas extraction industry workers based in the United States who meet the inclusion criteria qualifications were allowed to access Centiment's dashboard and complete the online survey.

Upon opening the survey within Centiment, the recruited participants were presented with the informed consent document that they were required to agree to before they could participate in the survey. In the event the participants agreed, they were permitted to complete the survey. A participant who selected no was thanked for their time and willingness to participate, the survey was stopped, and no additional data was collected.

Centiment recorded and stored participants' survey responses and provided the researcher with the results of the survey. The initial survey window was one week. Responses were collected starting April 18, 2023. The survey had to be extended until May 5, 2023 in order to

collect the number of responses desired for the study. Centiment used an exclusive tagging system to assign a unique identifier to each participant as they entered the survey, so names and other personally identifiable information were not disclosed. As part of Centiment's data collection process, transport layer security (TLS) was used to encrypt the data prior to transmitting the information to the researcher for analysis. The researcher will store the data on an encrypted storage drive for at least three years. Once three years have passed, the researcher will then delete the data.

Analytical Methods

This study utilized a correlational research design to examine the relationship between the variables of the study. The researcher used the R Commander and JASP statistics program to run correlation tests for this study. JASP and the R Commander ultimately result in the same data. Both applications that provide a graphical user interface (GUI) for R, an open-source, free statistical software (Sardareh et al., 2021).

The measurement of the strength of the relationship between variables is defined as a correlation (Oleson et al., 2022). Using survey data and correlational testing data analysis procedures, the researcher examined the relationships using a quantitative, non-experimental methodology.

Descriptive Analysis

The main objective of this study was to detect the presence and strength of a relationship between variables. The data collected was analyzed using descriptive and inferential analysis. To explain the data collected from the sample, descriptive statistics of the data set must be summarized, and the central tendency and measures of dispersion of the data must be identified (Mishra et al., 2019). The descriptive analysis allowed the researcher to look for trends, describe

the data set, and give information about the population. The descriptive analysis included the average, mean, standard deviation, and minimum and maximum for the available parameters of the study.

Inferential Analysis

According to Wonu et al. (2018), correlations can validate a research hypothesis. Spearman rho statistical are used to test allegations about the existence of a relationship between variables in an ordinal database that do not meet the assumption of normality (Wonu et al., 2018). Due to the normality scores and presence of outliers in the dataset of this study, a Spearman rho correlation was conducted on the dataset. In order to calculate the Spearman rho correlation coefficient, it is assumed that the joint distribution of the variables is normal (Janse et al., 2021). Rather than using the actual values of observations, Spearman's correlation coefficient uses the rank of observations when ordering observations from small to large (Fitriadi et al., 2022). As a result of this method of using the rank, it is robust in the face of outliers. Graphs and statistics were used to address the assumption of normality for the data analysis of this study using a predetermined Type I alpha error rate of .05 for hypothesis testing.

Ethical Considerations

The institutional review board (IRB) of Columbia Southern University reviewed and approved this study before it commenced. No data was collected, and no respondents were contacted until official IRB approval was obtained. Centiment obtained informed consent and applied confidentiality measures to protect the privacy of participants in the process of administering the questionnaires and survey for this study. The data was gathered using convenience sampling.

The National Research Act of 1974 mandated that the National Commission for the Protection of Human Subjects of Biomedical and Behavioral Research be created to develop guidelines that ensure biomedical and behavioral research involving human subjects is conducted in accordance with basic ethical principles (Office for Human Research Protections, 2022). The commission was also charged with developing guidelines for ensuring such research is conducted in compliance with those principles and the Belmont Report was created to publish the guidelines.

The Belmont Report's includes the requirements for the respect of peoples' personally identifiable information (PII) (Office for Human Research Protections, 2022). Collected data that includes PII about survey participants will be maintained in an anonymous manner. Furthermore, in accordance with the Belmont Report's principle for justice, participants for a research study must be selected fairly.

Additional measures were taken to ensure data collection was ethically completed. Coercion was not used to sway survey participation, participants had no relationship with the researchers, and no population was specifically targeted except to the extent to which respondents were required to meet the inclusion criteria. The research participants were selected using purposive sampling to ensure that those who meet the inclusion criteria for the study were permitted to participate. The secured and protected data was then provided in Excel format to the researcher for analysis. To ensure the confidentiality of the survey data and the privacy of the survey participant, the data will be stored securely for a minimum of three years and then destroyed by the researcher.

Summary

An examination of the relationship between transformational leadership, occupational safety climate, and job satisfaction was carried out in this quantitative, non-experimental, correlational study. The study's population was oil drilling and gas extraction industry workers located in the United States. The study's research question and hypotheses, as well as the research design and methodology, were outlined in Chapter 3. Furthermore, the study's purpose and problem statement were reiterated. Moreover, a description of the target population, instruments, validity and reliability, and the methods of data collection and analysis were provided. Finally, this chapter concluded with a discussion of the study's ethical considerations and limitations.

Chapter 4 includes a description of the study's data analysis process. As an introduction to the content of the chapter, a summary of the data analysis procedures described in Chapter 3, as well as the exploratory data analysis conducted as a part of this study, will be discussed. Discussion about descriptive data, including the sample population, characteristics, and demographics of the study's participants, is also included. The final section of Chapter 4 provides a detailed explanation of how the data was analyzed, any possible errors that may have occurred, and the final results of the hypothesis testing.

CHAPTER 4: DATA ANALYSIS AND RESULTS

The purpose of this quantitative correlational study was to examine the relationship between transformational leadership, occupational safety climate, and job satisfaction of US oil drilling and gas extraction industry workers. The research design utilized for this study was a non-experimental correlational approach that allowed the researcher to study constructs that are not directly observable, which allowed the researcher to determine the interaction between the study's variables without inferring causation. The purpose of this study was to address the following research question: What is the relationship between transformational leadership, occupational safety climate, and job satisfaction of oil drilling and gas extraction industry workers in the United States? Centiment was used to recruit participants and distribute the survey.

Data Analysis Procedures

The main objective of this study was to detect the presence and strength of a relationship between variables. In order to achieve adequate statistical power, a sample size of at least 102 respondents was required as determined by a G*Power analysis. One hundred and seven responses were received upon completion of the survey window. In order to participate in the study, individuals were required to be at least 18 years old and work in the oil drilling and gas extraction industry. The method of data collection was an online survey.

The data was analyzed using descriptive and inferential analysis. The data set was summarized descriptively, and the central tendency and measures of dispersion were identified. The descriptive analysis allowed the researcher to look for trends, describe the data set, and give information about the population. The descriptive analysis also included the average, mean, standard deviation, and minimum and maximum for the available parameters of the study.

To examine the relationship between the variables in this study, a correlational research design was used. A quantitative, non-experimental methodology to examine the relationships based on survey data and correlational testing data analysis. The researcher utilized the JASP statistics program to run correlation tests. In this study, Spearman rho was used to determine the existence of a relationship between variables due to the presence of normality scores and outliers in the dataset. This study used a predetermined Type I alpha error rate of .025 for hypothesis testing due to each variable being used twice which required the application of the Bonferroni correction In multiple statistical tests, Bonferroni corrections adjust probability (p) values because there is an increased risk of making type I errors (Armstrong, 2014).

Descriptive Data Analysis

Overall, 107 participants met the inclusion criteria for the study and completed the survey, which exceeded the minimum of 102. The G*Power (Figure 5) post hoc data indicated that the power of the sample was 0.82, which was over the 0.80 required power (Figure 4). Tests were also conducted to determine the validity of questions from MLQ 5x-short, NOSACQ-50, and JSS surveys. The reliability tests were carried out using the Rcommander scale reliability test.

Figure 5

Post hoc G*Power Analysis



The MLQ 5x-short was found to be internally consistent and reliable, with an alpha of 0.91. Separately, the transformational leadership survey results were tested for internal consistency and reliability and the Cronbach's alpha was 0.91. Next, the NOSACQ-50 survey was determined to be internally consistent and reliable with a Cronbach's alpha score of 0.96. Finally, the JSS survey was found to be internally consistent and reliable, with a Cronbach's alpha reliability score of 0.95. The positive results from the post hoc computed power and survey reliability allowed the researcher to move forward in generating sample demographics and descriptive statistics for the study.

Table 4

Cronbach's Alpha Reliability Analysis

	N	# of items	Cronbach's a	
MLQ 5x-short	107	45	.91	
NOSACQ-50	107	50	.96	
JSS	107	36	.95	

Descriptive Statistics

The study sample consisted of 107 participants who work in the oil drilling and gas extraction industry and live in the United States. Table 4 presents the demographic statistics of the collected survey data. Of the 107 participants, 75% (n = 80) were male and 25% were female (n = 27). The White or Caucasian race represented the largest percentage of survey participants at 81% (n = 87) followed by Hispanic (7%, n = 7), other race/ethnicity (6%, n = 6), and Black or African American (5%, n = 5). American Indian or Alaska Native and Asian participants each made up 1% (n = 1) of the sample.

Table 5

Demographic Statistics	Full	Sample
-	п	%
Gender		
Female	27	25
Male	80	75
Age		
18-29	5	5
30-44	68	64
45-59	25	24
60 or older	9	8
Education		
Bachelor's Degree	56	52
High School Diploma or Equivalent	29	27
Master's Degree	18	17

Demographic Statistics from Survey Data Set

Some High School	1	1
Terminal Degree or Higher	3	3
Race/Ethnicity		
American Indian or Alaska Native	1	1
Asian	1	1
Black or African American	5	5
Hispanic	7	7
Other	6	6
White or Caucasian	87	81
Managerial Position		
Yes	35	33
No	72	67

Sixty-four percent (n = 68) of participants were 30-44 years old, 24% (n=25) were 45-59

years old, 8% (n = 9) were 60 years, and 5% (n = 5) were 18 to 29 years old. The largest education representation of the study were participants with bachelor's degrees (52%, n = 56). Participants with high school diploma or equivalent represented 27% (n = 29) of the sample followed by 17% of participants (n = 18) with a master's degree. Terminal degree holding participants represented 3% of the sample (n = 3). Only 1% (n = 1) of participants had less than a high school diploma or equivalent.

Results

Below is a summary of the data.

Table 6

Numerical Summaries of Study Variable Groups

	M	SD	Skew	Kurt	п
Job Satisfaction	150.112	31.533	.098	-0.371	107
Transformational Leadership	3.681	.612	440	.480	107
Occupational Safety Climate	3.207	.506	.0322	-1.304	107

The researcher used JASP software to generate a histogram, boxplot, and QQ plot of the MLQ 5x-short, NOSACQ-50, and JSS variables, based on the descriptive statistics of the dataset. The dataset included the numerical summaries of the variable averages. These graphs facilitate the understanding of the data set and enables the assumption that data for the NOSACQ-50 and JSS does not meet the assumption of normality and that these data sets are not normally distributed. The graphs also facilitate the understanding of the dataset and allows for the assumption that the dataset does not represent a normal distribution. Table 7 shows the Anderson-Darling normality tests results for the study variables.

Figure 6



MLQ 5x-short, NOSACQ-50, and JSS Histograms

Figure 7





Note. These figures represent a boxplot for each of the variable in this study. The MLQ boxplot indicates that there is one outlier.

Figure 8



MLQ 5x-short, NOSACQ-50, JSS QQ Plots

Note. Frequency scores for the study variables. Outliers noted. Data points within the shaded area represent a 95% confidence. The NOSACQ-50 and JSS show a tendency toward not normally distributed data. The MLQ 5x-short, NOSACQ-50 and JSS QQ plots all indicate there are outliers.
Figure 9





Note. These figures represent scatterplots for each of the variables in this study. These scatterplots indicate that there is a linear relationship between the variables.

The MLQ 5x-short, NOSACQ-50 and JSS Scatterplots all indicate there are outliers.

Table 7

0	Anderson-Darling	p-value	
MLQ 5x- short	.586	.125	
NOSACQ-50	1.923	<.001	
JSS	1.170	.004	

Anderson-Darling Normality Test

The results indicate that the MLQ 5x- short data set is normally distributed. The results of the NOSACQ-50 and JSS the data sets are not normally distributed; thus, a Spearman rank test was used to test the relationship between the study's variables. A Pearson's r correlation could have been run on the MLQ 5x- short; however, the researcher opted to utilize the same correlation test, the Spearman rho, for all correlation tests for this study. Outliers found in the survey results were investigated by the researcher. Participants may experience disinterest when completing surveys because online survey data collection is relatively easy (Singh & Sagar, 2021). Other potential causes of outliers are that the participant may respond casually to some or all survey questions due to survey fatigue. Overall, due to the outliers found in the study results, the researcher utilized the Spearman rho correlation test to analyze the data.

The Spearman rank correlation was computed to assess the relationship between transformational leadership and job satisfaction and the results can be seen in Table 5. There was a positive relationship between the two variables, r(105) = .252, p= value .009. Therefore, the H1₀ hypothesis was rejected and the H1_a hypothesis was accepted. A positive relationship between transformational leadership and job satisfaction exists in oil drilling and gas extraction workers in the United States.

Next, the Spearman rank correlation was computed to assess the relationship between occupational safety climate and job satisfaction. There is a positive relationship between the two variables, r(105) = .669, p= value <.001. This resulted in the rejection of the null hypothesis and in the acceptance of the H2_a hypothesis. A positive relationship between occupational safety climate and job satisfaction exists in oil drilling and gas extraction workers in the United States.

Finally, the Spearman rho correlation test was run to test the relationship between transformational leadership and occupational safety climate. The results indicated indicates a

positive relationship between the two variables, r(105) = .364, p= value <.001. This resulted in the rejection of the null hypothesis and acceptance of the H3_a hypothesis. Thus, indicating a positive relationship between transformational leadership and occupational safety climate exists in oil drilling and gas extraction workers in the United States.

Table 6

Spearman Rho Correlation Coefficients for the MLQ 5x-short, NOSACQ-50, & JSS

	n	r(df)	p-value	decision
TL & JS	107	.252	.009**	Reject the Null
OSC & JS	107	.699	<.001**	Reject the Null
TL & OSC	107	.364	<.001**	Reject the Null

Note. ** Indicates a statistically significant p-value.

Summary

A quantitative non-experimental research design approach utilizing a Spearman rho correlation testing method was used to examine the relationship between transformational leadership, occupational safety climate, and job satisfaction of oil drilling and gas extraction industry workers in the United States. The study sample consisted of 107 participants who worked in oil drilling and gas extraction industry, were at least 18 years old, and lived in the United States.

The Spearman rho correlation coefficient for the relationship between transformational leadership and job satisfaction is .252 with a corresponding p-value of .009, which indicates a significant positive relationship between the variables and resulted in the rejection of the null hypothesis. The Spearman rho correlation for the relationship between occupational safety climate and job satisfaction is .699 with a corresponding p-value of <.001, which indicates a significant positive relationship between the variables and resulted in the rejection of the null

hypothesis. Finally, the Spearman's rho correlation coefficient for the relationship between transformational leadership and occupational safety climate is .364 with a corresponding p-value of <.001, which indicates a significant positive relationship between the variables and resulted in the rejection of the null hypothesis.

The significance of the study, its contributions to learning, and its connections to organizational commitment are covered in detail in Chapter 5 along with a summary of the research, goals, and an explanation of how the research question were addressed. The researcher will summarize the major conclusions in Chapter 5 and interpret the theoretical and practical implications of the conclusions in relation to the research question and hypotheses. The researcher also outlines suggestions for further study into organizational commitment and learning levels. The value and significance of this study are discussed in the conclusion of the chapter.

CHAPTER 5: SUMMARY, CONCLUSIONS, AND RECOMMENDATIONS

In many organizations, safety culture is ignored completely, which can lead to diminished job satisfaction (Niartiningsih et al., 2020). This research adds knowledge to the field of study by demonstrating the importance of the occupational safety climate in organizations and the relationship transformational leadership has with occupational safety climate and job satisfaction. Among the major concerns of business leaders are job satisfaction and safety for their workers (Sánchez-Segura et al., 2021). When business leaders better understand the relationship between occupational safety climate and transformational leadership, they can motivate followers to develop their abilities, which can increase organizational performance and improve organizational climate. Oil drilling and gas extraction industry leaders can use the results of this study to enhance their workplaces if they understand how the relationship between these variables may affect their organizations.

Chapter 1 includes an introduction to this study and a discussion of the background, problem statement, purpose statement, research questions, and hypotheses. Information on advancing theoretical knowledge, the significance of the study, and the study's application to business management was also presented. Chapter 1 concludes with the study's methodology, research design, term definitions, assumptions, limitations, and delimitations. Chapter 2 comprises a literature review of the variables, theories, frameworks, and instruments used to complete the data collection for this study. Chapter 3 contains detailed information on the study methodology. In addition, this chapter describes the techniques used in the study to allow future researchers to duplicate the process. Chapter 4 includes the data collection process and analysis of this study. Actual findings of the study, analysis and evaluation of the results as applied to

real-world business relevance are presented. Finally, Chapter 5 includes the study's implications, recommendations, and conclusions.

Summary of the Study

To conduct an examination of the relationship between transformational leadership, occupational safety climate, and job satisfaction was the purpose of this qualitative correlational study. Participants were recruited by Centiment from the target population so the researcher could examine the relationship between the study's variables. One hundred and seven of the recruited participants met the inclusion criteria and completed the MLQ 5x-short, the NOSACQ-50, and the JSS. The researcher used Spearman rho correlation due to the data not being normally distributed. R commander and JASP software were utilized to analyze the collected data and created the tables and graphs for the study.

Summary of Findings and Conclusion

The aim of this study was to answer the following research question: What is the relationship between transformational leadership, occupational safety climate, and job satisfaction of oil drilling and gas extraction industry workers in the United States? The results of the data analysis indicate positive and statistically significant relationships between transformational leadership and job satisfaction, transformational leadership and occupational safety climate, and transformational leadership and occupational safety climate among oil drilling and gas extraction workers aged 18 years and older in the United States.

The following section summarizes the study's theoretical implications. The results of this study are further explained in relation to the theory of transformational leadership proposed by Bernard Bass, Heinrich's domino theory, and Maslow's hierarchy of needs. Lastly, how the

findings can directly impact businesses and business leaders in their professional practice will be examined and discussed.

Implications

This study provides information to business leaders at all levels. The data obtained in this study can be used to improve leadership styles, safety culture, and a positive work environment. These changes can ultimately result in better job satisfaction for employees. This section will cover the theoretical implications consistent with the theoretical framework presented in Chapter 2. Professional practice implications of this study will also be discussed.

Theoretical Implications

Bernard Bass' transformational leadership theory, H.W. Heinrich's domino theory, and Abraham Maslow's hierarchy of needs were used as the theoretical frameworks for this study. By understanding the theories in this study, leaders can understand the best ways to influence their employees. This influence can lead to creation of a work environment that is conducive to both job satisfaction and safety. Such an environment could foster collaboration, creativity, and shared responsibility. These environmental characteristics can be achieved by leveraging the positive effects of transformational leadership, occupational safety climate, and job satisfaction (Asgari et al., 2020).

To best understand how the study results relate to transformational leadership, domino, and hierarchy of needs theories, it is important to summarize any theoretical implications that may be derived from the results. Transformational leadership theory posits that a transformational leader can inspire and motivate employees to strive for excellence. Employees who are inspired and striving for excellence can identify and report the initial hazard, which will result in speedy remediation and avoidance of workplace mishaps and injuries.

Heinrichs domino theory places the focus of accidents on unsafe behavior. When the most influential accident prevention theory in occupational safety and health discipline is combined with the benefits that a transformational leader provides, employees are inspired, excellent, happy, and dedicated to the organization (Dunlap et al., 2019; Yorio & Moore, 2018). Creating a culture that builds up employees who are striving for excellence results in a safe work environment (Judge, 2018). Such a culture creates employees who will be inspired by transformational leaders, which can help reduce human error and unsafe acts in the workplace (Yorio & Moore, 2018).

Finally, the hierarchy of needs theory establishes the basis for employee motivation (Fallatah & Syed, 2018). When employees have a safe and healthy work environment, the foundational layer of Maslow's hierarchy of needs theory is satisfied (Fallatah & Syed, 2018). Satisfied, safe, and secure employees are motivated and productive, and this leads to higher customer loyalty, profitability, and production and lower turnover for an organization (DeLeon, 2023).

Implications for Professional Practice

This study adds to the knowledge of limited research on the relationship between transformational leadership, occupational safety climate, and job satisfaction. The data that was collected from employees in the oil drilling and gas extraction industry in the United States show that job satisfaction is related to transformational leadership and occupational safety climate. The results of this study can have many implications for professional practice. For example, because transformational leaders are known to motivate and inspire their followers and nurture a safe work environment, employees tend to be more satisfied and committed to their organizations.

This study provides implications for an organization's business leaders and human resources management teams. These implications are due to the relationship of the study's variables and the impact this relationship can have on the organization. Employee morale, engagement, and retention are some of the most difficult challenges for organizations to overcome (DeLeon, 2023). Productivity, employee turnover and retention, injury cost, accident and mishap prevention, and organizational and employee performance rates are all affected by these relationships. Transformational leadership and a positive occupational safety climate can contribute to increased productivity, motivation, and job satisfaction and reduced injury rates (Akbari et al., 2019; Akdere & Egan, 2020).

Business leaders and human resource management teams can also tap into new knowledge and use the information presented in the study as a resource to develop strategies that encourage employees to be creative, as transformational leaders inspire this trait in their employees (Akdere & Egan, 2020). Businesses should use this to challenge assumptions and their employees to find new and better ways to conduct business. Finally, the connection between occupational safety climate and job satisfaction is a safer work environment and happier employees who contribute to increased organizational performance.

Recommendations

All research is delimited to ensure timely and successful completion of a study. Therefore, it is important to identify potential future research opportunities and outline ways that the current research can be applied by businesses. Recommendations for future research and for practice are outlined in the following sections.

Recommendations for Future Research

Transformational leadership, occupational safety climate, and job satisfaction play an intricate role in an organization's culture. There is a statically significant relationship between the variables for the sample of oil drilling and gas extraction workers in the United States that was surveyed. Additional qualitative research should be conducted to see if the results can be replicated in different geographic locations. This type of research would allow the researcher to determine if specific geographical social and cultural factors impact the results of the study. One recommended geographic location is the Middle East, where there is a large oil drilling and gas extraction industry.

Each additional research study conducted taps into potential new knowledge on the relationship between transformational leadership, occupational safety climate, and job satisfaction. Further research is needed to unravel the extent of the impact this relationship has on an organization and its employees. Additional research on the relationship between transformational leadership, job satisfaction, and occupational safety climate should be conducted utilizing a different type of quantitative research, such as experimental.

Additional investigation into the complex interplay between these three variables can help organizational leaders better understand how to foster a safe and fulfilling work environment. This study utilized a Likert-based survey that was designed to capture employee perception about their organizations. Closed-ended questions cannot elicit understanding of feelings beyond what the scale can measure. Therefore, a follow-up qualitative study should be conducted to gain insight into how employees truly feel about their organizations. This can be accomplished through the use of interviews or focus groups. Further research could also provide valuable insight into how to effectively motivate employees to prioritize workplace safety and

create a safer work environment and inform organizations about how to implement strategies that create better working conditions. Results from this type of study would likely allow business leaders to make changes that would improve job satisfaction and increase employee retention.

This study's target population was limited to oil drilling and gas extraction industry workers in the United States. Future research should expand into other industries. This research would allow researchers to gain more insight into the relationship between the study's variables and to identify if the type of industry affects how employees view their organizations in general. This further research may help to uncover potential patterns of leadership and safety cultural behaviors present in different industry across the globe.

Recommendations for Practice

The data obtained in this study cannot be used to identify the leadership characteristics present in this study's sample because the questions were geared toward how the participants view their leaders, not their own leadership styles. Additionally, the data collected during this study only reveal how these specific employees feel about their leaders, their perception of their organization's safety climate, and their job satisfaction. A safety culture that can lead to improved employee job satisfaction can be developed based on the results of this study. Transformational leadership encourages employees to think critically and take initiative, while a positive occupational safety climate promotes trust and a sense of safety in the workplace. Strong job satisfaction can lead to improved productivity and, when combined with transformational leadership and a positive safety climate, collaboration and creativity within a team can be strengthened.

It is important to create a work environment that meets employees' needs and encourages self-actualization. Organizations with work environments that promote these characteristics may

find that employee commitment is nurtured, employee retention is increased, and a safe and productive workplace environment is created. Transformational leaders promote their employees' needs and ensure that safety conditions are adequate, which fosters and increases employee job satisfaction, ultimately resulting in increased commitment and loyalty. Employees who are committed to their organizations are more productive, willing to learn, develop more skills, and perform better (Matsuo, 2018).

Concluding Remarks

The purpose of this quantitative correlational study was to examine the relationship between transformational leadership, occupational safety climate, and job satisfaction of US oil drilling and gas extraction industry workers. An exhaustive review of the literature revealed no studies that examined these variables in that industry. Thus, examining the relationship between these three variables expanded the existing body of knowledge. An analysis of the data indicates that there is a statistically significant relationship between transformational leadership, occupational safety climate, and job satisfaction among the target population. This research provides an in-depth understanding of how the perception of an organization's safety culture and leadership can impact employees' job satisfaction.

The results of this study add valuable information to existing literature that can be used by organizational leaders, trainers, safety professionals, and human resources teams to develop strategies for training impactful leaders and initiatives that could benefit the organization in the long term, which could contribute to increased job satisfaction. Business leaders and human resource management teams can use this information to incorporate transformational leadership training into their organization. These leaders and team members can also use this information to

develop safety programs within their organizations that will fulfill employees' basic safety needs.

Understanding the dynamics of the relationship between leadership and employee safety can help organizations create a work environment that is more conducive to productivity and employee engagement and wellbeing (Shabane et al., 2017; Thompson et al., 2021; Waldman et al., 2012; Willis et al., 2017). The results of this study indicate that the variables are statistically interconnected, providing evidence of the importance of leadership in creating an environment for employees that is both safe and satisfying. Employee satisfaction and safety is good for the organization as well and is worth pursuing. In conclusion, the following words from the famous and gifted author, Simon Sinek (2014), ring as true today as they did when he first said them: "When a leader makes the choice to put the safety and lives of the people inside the organization first, to sacrifice their comforts and sacrifice the tangible results, so that the people remain and feel safe and feel like they belong, remarkable things happen" (4:39).

REFERENCES

- Abbas, Q., & Khanam, S. J. (2020). Adaptation and psychometric properties of Urdu version of job satisfaction survey. *Journal of Behavioral Science*, *9*, 57-74.
- Akbari, H., Kashani, M. M., Asadi, Z., Kaveh, M., & Saberi, H. (2019). The relationship between job satisfaction and the incidence of unsafe acts in metal smelting industry workers in 2017. *International Archives of Health Sciences*, 6(3), 127. https://doi.org/10.4103/iahs.iahs 40 19
- Akdere, M., & Egan, T. (2020). Transformational leadership and human resource development:
 Linking employee learning, job satisfaction, and organizational performance. *Human Resource Development Quarterly*, 31(4), 393–421. <u>https://doi.org/10.1002/hrdq.21404</u>
- AL-Nawafah, S. S., Al-Amaeara, A. F., & Alkhawaldah, R. A. (2022). Applying transformational leadership in Jordan governmental institutions. *Journal of Management Information & Decision Sciences*, 25, 1–13. https://www.researchgate.net/publication/354778551
- Alimo-Metcalfe, B., & Alban-Metcalfe, R. J. (2001). The development of a new transformational leadership questionnaire. *Journal of Occupational and Organizational Psychology*, 74(1), 1–27. https://doi.org/10.1348/096317901167208
- Amalina, N. N., Armanu, & Susilowati, C. (2022). The effect of transformational leadership and organizational justice toward employee performance mediated by job satisfaction: A study of Blitar, Indonesia police. *International Journal of Research in Business & Social Science*, 11(4), 95–105. <u>https://doi.org/10.20525/ijrbs.v11i4.1771</u>

- Andrade, C. (2018). Internal, external, and ecological validity in research design, conduct, and evaluation. *Indian Journal of Psychological Medicine*, 40(5), 498–499.
 https://doi.org/10.4103/ijpsym.ijpsym_334_18
- Armstrong, R. A. (2014). When to use the Bonferroni correction. *The Journal of the College of Optometrists*, 34, 502-508. https://doi.org/10.1111/opo.12131

Asgari, A., Mezginejad, S., & Taherpour, F. (2020). The role of leadership styles in organizational citizenship behavior through mediation of perceived organizational support and job satisfaction. *Innovar*, 30(75), 87–98.
https://doi.org/10.15446/innovar.v30n75.83259

- Avolio, B., & Bass, B. (2004). *Multifactor leadership questionnaire* (3rd ed.). Mind Garden. https://www.mindgarden.com/16-multifactor-leadership-questionnaire
- Badura, K. L., Galvin, B. M., & Lee, M. Y. (2021). Leadership emergence: An integrative review. Journal of Applied Psychology. <u>https://doi.org/10.1037/ap10000997.supp</u>
- Barbinta, A., Dan, S., & Muresan, C. (2017). Bernard Bass founder of the transformational leadership theory. *Review of Management & Economic Engineering*, 16(4), 758–762. http://www.rmee.org/abstracturi/66/22
- Barling, J., Loughlin, C., & Kelloway, E. K. (2002). Development and test of a model linking safety-specific transformational leadership and occupational safety. *Journal of Applied Psychology*, 87(3), 488–496. <u>https://doi.org/10.1037/0021-9010.87.3.488</u>
- Bass, B. M. (1985). Leadership: Good, better, best. Organizational Dynamics, 13(3), 26–40. https://doi.org/10.1016/0090-2616(85)90028-2

- Bass, B. M., Avolio, B. J., Jung, D. I., & Berson, Y. (2003). Predicting unit performance by assessing transformational and transactional leadership. *Journal of Applied Psychology*, 88(2), 207–218. <u>https://doi.org/10.1037/0021-9010.88.2.207</u>
- Batista-Foguet, J. M., Esteve, M., & van Witteloostuijn, A. (2021). Measuring leadership an assessment of the multifactor leadership questionnaire. *PLOS ONE*, 16(7). https://doi.org/10.1371/journal.pone.0254329
- Batura, N., Skordis-Worrall, J., Thapa, R., Basnyat, R., & Morrison, J. (2016). Is the job satisfaction survey a good tool to measure job satisfaction amongst health workers in Nepal? Results of a validation analysis. *BMC Health Services Research*, 16(1). https://doi.org/10.1186/s12913-016-1558-4
- Berthelsen, H., Muhonen, T., Bergström, G., Westerlund, H., & Dollard, M. F. (2020).
 Benchmarks for evidence-based risk assessment with the Swedish version of the 4-Item psychosocial safety climate scale. *International Journal of Environmental Research and Public Health*, 17(22). <u>https://doi.org/10.3390/ijerph17228675</u>
- Beus, J. M., Payne, S. C., Arthur, W., & Muñoz, G. J. (2017). The development and validation of a cross-industry safety climate measure: Resolving conceptual and operational issues. *Journal of Management*, 45(5), 1987–2013. <u>https://doi.org/10.1177/0149206317745596</u>
- Bolle, J. (2022, August 2022). Using the accident triangle to grow your safety culture. *Xecellerate*. <u>https://www.excelleratemfg.com/blog</u>
- Borell, D. A., Camfield, C. E., & Rodrigues, R. G. (2019). An analysis of the relationship between personal values and satisfaction with work. *Brazilian Journal of Management/ Revista De Administração Da UFSM.*, 11(5), 1240–1256. https://doi.org/10.5902/1983465916992

- Bowling, N., & Zelazny, L. (2021). Measuring general job satisfaction: Which is more construct valid—global scales or facet-composite scales? *Journal of Business and Psychology*, 37(1), 91–105. <u>https://doi.org/10.1007/s10869-021-09739-2</u>
- Brown, J. C. (2020). Web-based learning and non-traditional students: A quantitative causalcomparative study (Doctoral dissertation). Dissertations & Theses Full Text: The Humanities and Social Sciences Collection database. (Dissertation No. 28085949).
- Calis, C., & Buyukakinci, B. Y. (2019). Leadership approach in occupational safety: Taiwan sample. *Procedia Computer Science*, 158, 1052–1057. https://doi.org/10.1016/j.procs.2019.09.146
- CITI Program. (n.d.). Essentials of statical analysis (EOSA): Part 3, analysis of variance. https://www.citiprogram.org/members/index/cfm?pageID=665&ce=1#view
- Day, S. W., Lawong, D., Miles, A. K., & Effon, T. (2022). Leadership and culture in Ghana's tourism and hospitality industry: The impact of transformational leadership on job satisfaction in an emerging economy. *Journal of Leadership, Accountability & Ethics*, 19(1), 127–134.
- DeLeon, H. (2023, January 17). *How to increase employee job satisfaction*. Career Charge. https://corporatetraining.usf.edu/blog
- Dollard, M. F., Dormann, C., & Idris, M. A. (2019). *Psychosocial safety climate: A new work stress theory*. Springer.
- Draghici, A., Dursun, S., Başol, O., Boatca, M. E., & Gaureanu, A. (2022). The mediating role of safety climate in the relationship between transformational safety leadership and safe behavior—The case of two companies in Turkey and Romania. *Sustainability*, *14*(4).
 <u>https://doi.org/10.3390/su14148464</u>

- Dunlap, E. S., Basford, B., & DePoy Smith, M. L. (2019, May). Remodeling Heinrich: An application for modern safety management. *PSJ Professional Safety*, 64(5), 44–52.
- Eliyana, A., & Ma'arif, S. (2019). Job satisfaction and organizational commitment effect in the transformational leadership towards employee performance. *European Research on Management and Business Economics*, 25(3), 144-150.
- Fallatah, R. H., & Syed, J. (2018). A critical review of Maslow's hierarchy of needs. *Employee Motivation in Saudi Arabia*, 19–59. <u>https://doi.org/10.1007/978-3-319-67741-5_2</u>
- Fitriadi, F., Jiuhardi, J., Busari, A., Ulfah, Y., Permadi, H., Kurniawan, E., & Darma, D. (2022).
 Using correlation analysis to examine the impact of COVID-19 pandemics on various socioeconomic aspects: Case study of Indonesia. *Geographica Pannonica*, 26(2), 128–141. <u>https://doi.org/10.5937/gp26-37049</u>
- Flatau-Harrison, H., Griffin, M. A., & Gagne, M. (2020). Trickling down: The impact of leaders on individual role clarity through safety climate strength across time. *Safety Science*, 121, 485-495. <u>https://doi.org/10.1037/mgr0000108</u>
- Free Workplace Safety Program Reduces Workers' Compensation Costs. (2022, July 12). ABNewswire, NA. <u>https://link-gale-com</u>
- Gaviria-Rivera, J. I., & Lopez-Zapata, E. (2019). Transformational leadership, organizational climate and job satisfaction in work teams. *European Research Studies*, *22*(3), 68.
- Grill, M., Nielsen, K., Grytnes, R., Pousette, A., & Törner, M. (2019). The leadership practices of construction site managers and their influence on occupational safety: An observational study of transformational and passive/avoidant leadership. *Construction Management & Economics*, 37(5), 278–293. <u>https://doi.org/10.1080/01446193</u>

- Guarnaccia, C., Scrima, F., Civilleri, A., & Salerno, L. (2016). The role of occupational selfefficacy in mediating the effect of job insecurity on work engagement, satisfaction, and general health. *Current Psychology*, 37(3), 488–497. <u>https://doi.org/10.1007/s12144-016-</u> 9525-0
- Gyekye, S. A., & Haybatollahi, M. (2014). Relationship between organizational justice and organizational safety climate: Do fairness perceptions influence employee safety behaviour? *International Journal of Occupational Safety and Ergonomics*, 20(2), 199–211. https://doi.org/10.1080/10803548.2014.11077045
- Gyekye, S. A., & Saiminen, S. (2007). Workplace safety perceptions and perceived organizational support: Do supportive perceptions influence safety perceptions? *International Journal of Occupational Safety and Ergonomics*, 13(2), 189–200. <u>https://doi.org/10.1080/10803548.2007.11076721</u>
- Hall, M. E., Blair, E. H., Smith, S. M., & Gorski, J. D. (2013). Development of a theory-based safety climate instrument. *Journal of Safety, Health & Environmental Research*, 9(1), 50-69
- Inness, M., Turner, N., Barling, J., & Stride, C. B. (2010). Transformational leadership and employee safety performance: A within-person, between-jobs design. *Journal of Occupational Health Psychology*, 15(3), 279–290. <u>https://doi.org/10.1037/a0019380</u>
- Janse, R. J., Hoekstra, T., Jager, K. J., Zoccali, C., Tripepi, G., Dekker, F. W., & van Diepen, M. (2021). Conducting correlation analysis: important limitations and pitfalls. *Clinical Kidney Journal*, 14(11), 2332–2337. <u>https://doi.org/10.1093/ckj/sfab085</u>
- Jauhar, J., Ting, C. S., Rahim, A., & Abdul Rahim, N. F. (2017). The impact of reward and transformational leadership on the intention to quit of generation y employees in oil and

gas industry: Moderating role of job satisfaction. *Global Business & Management Research*, 9, 426–441.

- Judge, J. L. P. (2018). A qualitative and quantitative examination of factors underlying safety climate (Doctoral dissertation). Dissertations & Theses Full Text: Industrial and Agricultural Technology Collection database. (Dissertation No. 12318739).
- Kaluza, A. J., Schuh, S. C., Kern, M., Xin, K., & Dick, R. (2020). How do leaders' perceptions of organizational health climate shape employee exhaustion and engagement? Toward a cascading-effects model. *Human Resource Management*, 59(4), 359–377. https://doi.org/10.1002/hrm.22000
- Kasztelnik, K., & Al-ahmadi, A. (2022). The investigation of the transformational leadership style managers and impact on the financial managerial performance with the modern distributed graph analysis and parallel coordinates algorithm in the United States. *Journal of Applied Business and Economics*, 24(1), 1–16.
 https://doi.org/10.33423/jabe.v24i1.4945
- Kines, P., Lappalainen, J., Mikkelsen, K. L., Olsen, E., Pousette, A., Tharaldsen, J., Tómasson,
 K., & Törner, M. (2011). Nordic Safety Climate Questionnaire (NOSACQ-50): A new
 tool for diagnosing occupational safety climate. *International Journal of Industrial Ergonomics*, 41(6), 634–646. <u>https://doi.org/10.1016/j.ergon.2011.08.004</u>
- Kubasinski, S., & Slawinska, M. (2021). Research on corporate social responsibility (CRS) in terms of work safety. *European Research Studies*, 24(S5), 626+. https://link.gale.com/apps/doc/A689478167
- Kwang-rae, C., Park, M. K., & Lee Hong-Jae, L. (2016). The mediator effect of communication on relationship between transformational leadership and job performance: Focusing on

the cognition of police officers. *Korean Public Management Review*, 30(1), 1–25. https://doi.org/10.24210/kapm.2016.30.1.001

- Lal, H. (2022). Fast-tracking safety culture in industry or face incidents/losses. *Journal of Psychosocial Research*, 17(1), 143–169. https://doi.org/10.32381/jpr.2022.17.01.13
- Lee, Y. D., Chen, P., & Su, C. (2020). The evolution of the leadership theories and the analysis of new research trends. *International Journal of Organizational Innovation*, 12(3), 88– 104.
- Le Tourneau University. (2022, October 1). *Quantitative research and analysis: Quantitative methods overview*. https://lib-guides.letu.edu/quantresearch/quantmethods
- Limberg, D., Gnilka, P. B., & Broda, M. (2021). Advancing the counseling profession by examining relationships between variables. *Journal of Counseling & Development*, 99(2), 145–155. <u>https://doi.org/10.1002/jcad.12362</u>
- Lin, C. Y., Huang, C. K., Li, H. X., Chang, T. W., & Hsu, Y. C. (2021). Will they stay or leave? Interplay of organizational learning culture and workplace mindfulness on job satisfaction and turnover intentions. *Public Personnel Management*, 51(1), 24–47. https://doi.org/10.1177/0091026021991581
- Loh, M. Y., Dollard, M. F., McLinton, S. S., & Tuckey, M. R. (2021). How psychosocial safety climate (PSC) gets stronger over time: A first look at leadership and climate strength. *Journal of Occupational Health Psychology*, 26(6), 522–536. https://doi.org/10.1037/ocp0000308
- Lyubykh, Z., Turner, N., Hershcovis, M. S., & Deng, C. (2022). A meta-analysis of leadership and workplace safety: Examining relative importance, contextual contingencies, and

methodological moderators. *Journal of Applied Psychology*, *107*(12), 2149–2175. https://doi.org/10.1037/ap10000557

Marquardt, N., Hoebel, M., & Lud, D. (2020). Safety culture transformation—The impact of training on explicit and implicit safety attitudes. *Human Factors and Ergonomics in Manufacturing & Service Industries*, 31(2), 191–207. <u>https://doi.org/10.1002/hfm.20879</u>

Matsuo, M. (2018). Goal orientation, critical reflection, and unlearning: An individual-level study. *Human Resource Development Quarterly*, 29(1), 49–66. <u>https://doiorg/10.1002/hrdq.21303</u>

- McCallaghan, S., Jackson, L. T., & Heyns, M. M. (2019). Transformational leadership, diversity climate, and job satisfaction in selected south African companies. *Journal of Psychology in Africa*, 29(3), 195–202. <u>https://doi.org/10.1080/14330237.2019.1619994</u>
- McCleskey, J. A., & Ruddell, L. (2020). Taking a step back—Maslow's theory of motivation: A Christian critical perspective. *Journal of Biblical Integration in Business*, *23*(1), 6–16.
- Mcleod, S. (2023). *Maslow's hierarchy of needs theory*. Simply Psychology. https://www.simplypsychology.org/maslow.html
- Mirza, M. Z., & Isha, A. S. N. (2020). Safety-specific transformational leadership, safety climate and occupational accidents. *Journal of Applied Structural Modeling*, 4(2), 44-52. <u>https://doaj.org/article/2f81ed0f562944c988ff564867a9a6f3</u>
- Mishra, P., Pandey, C. M., Singh, U., Gupta, A., Sahu, C., & Keshri, A. (2019). Descriptive statistics and normality tests for statistical data. *Annals of Cardiac Anesthesia*, 22(1), 67–72. <u>https://doi.org/10.4103</u>
- Muchiri, M. K., McMurray, A. J., Nkhoma, M., & Pham, H. C. (2019). How transformational and empowering leader behaviors enhance workplace safety: A review and research

agenda. The Journal of Developing Areas, 53(1), 257–265.

https://doi.org/10.1353/jda.2019.0015

- NFA.dk Det Nationale Forsknings Center for Arbejdsmiljø. (n.d.). Safety climate questionnaire-NOSACQ-50. <u>https://nfa.dk/da/Safety-Climate-Questionnaire-NOSACQ50</u>.
- Office for Human Research Protections. (2022, October 17). *The Belmont Report*. U.S. Department of Health & Human Services. <u>https://www.hhs.gov/ohrp/regulations-and-policy/belmont-report/index.html</u>
- Ojeda, M. J. (2023). *The examination of servant leadership style on employee empowerment and employee resistance* (Doctoral dissertation). Dissertations & Theses Full Text: The Humanities and Social Sciences Collection database. (Dissertation No. 29395102).
- Oleson, J. J., Jones, M. A., Jorgensen, E. J., & Wu, Y. H. (2022). Statistical considerations for analyzing ecological momentary assessment data. *Journal of Speech, Language, and Hearing Research*, 65(1), 344–360. <u>https://doi.org/10.1044/2021_jslhr-21-00081</u>

Occupational Safety and Health Administration. (1970). https://www.osha.gov/

- Osman, A., Khalid, K., & AlFqeeh, F. M. (2019, October 19). Exploring the role of safety culture factors towards safety behavior. *International Journal of Entrepreneurship*, 23(3). <u>https://www.abacademies.org/articles/Exploring-the-role-of-safety</u>
- Ozkan, A. H., Elci, M., Karabay, M. E., Kitapci, H., & Garip, C. (2020). Antecedents of turnover intention: A meta-analysis study in the United States. *E+M Ekonomie a Management*, 23(1). <u>https://doi.org/10.15240/tul/001/2020-1-007</u>
- Ramakrishnan, V. A. L., Bin, M., & Lasi, M. A. (2020). The influence of leadership on employee job performance among Royal Malaysian police in Klang Valley. *Asian*

Journal of Public Administration, 2(1), 1-11.

https://doi.org/10.18488/journal.138.2020.21.1.11

- Rapeli, L. (2022). What is the best proxy for political knowledge in surveys? *PLOS ONE*, *17*(8). https://doi.org/10.1371/journal.pone.0272530
- Renjith, V., Renu, G., & George, A. (2015). Transformational leadership in nursing. International Journal of Scientific Research and Management Studies, 2(2), 112–118.
- Reza, M. H. (2019). Components of transformational leadership behavior. *ERPA International Journal of Multidisciplinary Research*, 5(3), 119–124. https://www.researchgate.net/publication/333798276
- Roberts, C., & Hyatt, L. (2018). *The dissertation journey* (3rd ed.). SAGE Publications, Inc. (US). <u>https://bookshelf.vitalsource.com/books/9781506373331</u>
- Sapp, M. (2017). *Primer on effect sizes, simple research designs, and confidence intervals.* Charles C Thomas Publisher, Ltd.
- Sánchez-Segura, M. I., Dugarte-Peña, G. L., de Amescua, A., Medina-Domínguez, F., López-Almansa, E., & Reyes, E. B. (2021). Smart occupational health and safety for a digital era and its place in smart and sustainable cities. *Mathematical Biosciences and Engineering*, *18*(6), 8831–8856. <u>https://doi.org/10.3934/mbe.2021436</u>
- Sardareh, S. A., Brown, G. T. L., & Denny, P. (2021). Comparing four contemporary statistical software tools for introductory data science and statistics in the social sciences. *Teaching Statistics*, 43, S157–S172. <u>https://doi.org/10.1111/test.12274</u>
- Shabane, Z. W., Schultz, C. M., & Van Hoek, C. E. (2017). Transformational leadership as a mediator in the relationship between satisfaction with remuneration and the retention of

artisans in the military. *SA Journal of Human Resource Management*, *1*(2). <u>https://doi.org/10.4102/sajhrm.v15i0.923</u>

- Sinek, S. (2014, March). *Why good leaders make you feel safe* [Video]. TEDEd. https://ed.ted.com/lessons/why-good-leaders-make-you-feel-safe-simon-sinek
- Singh, S., & Sagar, R. (2021). A critical look at online survey or questionnaire-based research studies during COVID-19. Asian Journal of Psychiatry, 65, 102850. https://doi.org/10.1016/j.ajp.2021.102850
- Spector, P. E. (1985). Measurement of human service staff satisfaction: Development of the job satisfaction survey. *American Journal of Community Psychology*, *13*(6), 693–713.

https://doi.org/10.1007/bf00929796

- Spector, P. E. (1997). Job satisfaction: Application, assessment, cause, and consequences. SAGE Publications.
- Statistic Solutions. (n.d.). *Multifactor leadership questionnaire*. <u>https://www.statisticssolutions.com/</u>
- Steinmann, B., Klug, H. J., & Maier, G. W. (2018). The path is the goal: How transformational leaders enhance followers' job attitudes and proactive behavior. *Frontiers in Psychology*, 9, 1–15. <u>https://doi.org/10.3389/fpsyg.2018.02338</u>
- Summers, D., Sarris, A., Harries, J., & Kirby, N. (2022). The development of a brief and practical work safety climate measure. *International Journal of Industrial Ergonomics*, 87. <u>https://doi.org/10.1016/j.ergon.2021.103255</u>
- Niartiningsih, A., Irwandy, & Muis, M. (2020). The influence of occupational safety and health on job satisfaction and organizational commitment of hospital nurses in Makassar city.

Systematic Reviews in Pharmacy, 11(11), 1851–1855.

https://doi.org/10.37506/mlu.v20i3.1505

- Tan, L., Ma, Z., Huang, J., & Guo, G. (2021). Peer abusive supervision and third-party employee creativity from a social exchange theory perspective. *Social Behavior and Personality: An International Journal*, 49(5), 1–12. <u>https://doi.org/10.2224/sbp.9641</u>
- Theofanidis, D., & Fountouki, A. (2018). Limitations and delimitations in the research process. Perioperative Nursing, 7(3), 15-163. <u>https://doi.org/10.5281/zenodo.2552022</u>
- Thompson, G., Buch, R., Thompson, P. M. M., & Glasø, L. (2021). The impact of transformational leadership and interactional justice on follower performance and organizational commitment in a business context. *Journal of General Management*, 46(4), 274–283. <u>https://doi.org/10.1177/0306307020984579</u>
- Waldman, D. A., Carter, M. Z., & Hom, P. W. (2012). A multilevel investigation of leadership and turnover behavior. *Journal of Management*, 41(6), 1724–1744. <u>https://doi.org/10.1177/0149206312460679</u>
- Willis, S., Clarke, S., & O'Connor, E. (2017). Contextualizing leadership: Transformational leadership and management-by-exception-active in safety-critical contexts. *Journal of Occupational and Organizational Psychology*, 90(3), 281–305.
 https://doi.org/10.1111/joop.12172
- Wonu, N., Victor-Edema, U. A., & Ndimele, S. C. (2018). Test of significance of correlation coefficient in science and educational research. *International Journal of Mathematics and Statistics Studies*, 9(2), 53-68.

- Wu, M., Zhao, K., & Fils-Aime, F. (2022). Response rates of online surveys in published research: A meta-analysis. *Computers in Human Behaviors Reports*, 7. <u>https://doi.org/10.1016/j.chbr.2022.100206</u>
- Wu, T., Wang, Y., Ruan, R., & Zheng, J. (2022). Divergent effects of transformational leadership on safety compliance: A dual-path moderated mediation model. *PLOS ONE*, *17*(1), 1–17. <u>https://doi.org/10.1371/journal.pone.0262394</u>
- Yorio, P. L., & Moore, S. M. (2018). Examining factors that influence the existence of Heinrich's safety triangle using site-specific H&S data from more than 25,000 establishments. *Risk Analysis*, 38(4), 839–852. <u>https://doi.org/10.1111/risa.12869</u>

APPENDIX A

Informed Consent Agreement

You are invited to participate in a web-based online survey through Centiment. The research is being conducted by Amy Lynn Koehler-Warren, a Doctor of Business Administration student at Columbia Southern University. The survey is designed to examine the relationship between transformational leadership, occupational safety climate, and job satisfaction among oil drilling and gas extraction employees who are part of the United States population that are 18 years old or older and identify as male or female.

Your participation in this study is voluntary. It is completely up to you whether you participate in this study. If you decide to participate in this study, you will be asked to give electronic consent. After you have given consent, you are free to withdraw at any time and without giving a reason. Even after submitting the survey, you can still withdraw by emailing the researcher at <u>amylynnkoehlerwarren@gmail.com</u> and providing the full URL or aid value from when you completed the survey. Refusing to participate in the research or exiting the survey at any time will not result in any form of penalty.

The web-based online survey should take approximately 20-30 minutes to complete. Your responses will be confidential, and we do not collect identifying information such as name or email address. There are no foreseeable risks involved in participating in this study other than those encountered in day-to-day life. We will do our best to keep your information confidential. Your completed survey will be stored at Centiment, where all data is stored in a protected electronic format. The researcher will also store the raw data on an encrypted storage drive at their personal residence for a minimum of three years. To help protect your confidentially, the survey will not contain information that will personally identify you. The results of this study will be solely used for scholarly purposes only.

If you have questions about the study or the procedures at any time, you may contact the research supervisor, Dr. Tammy Corcoran, via email at

Tammy.Corcoran@columbiasouthern.edu .

ELECTRONIC CONSENT: Please select your choice below. You may print a copy of this consent form for your records. Clicking on the "Yes, I consent to participate in this study" button indicates that

- You have read the above information
- You voluntarily agree to participate
- You are 18 years of age or older

• Yes, I consent to participate in this study

• No, I do not consent to participate in this study

APPENDIX B

Survey Instrument

Demographic Questions

1. Are you male or female?

Male

Female

How old are you?
 Under 18 years old
 18-29 years old
 30-44 years old
 45-59 years old
 60 years or older

3. Are you an oil drilling and gas extraction industry worker?

Yes

No

4. What is your race/ethnicity?

Black or African American

White or Caucasian

Hispanic

American Indian or Alaska Native

Asian

Native Hawaiian or Other Pacific Islander

Other

5. What is the highest degree or level of education you have completed?

Some high school

High School diploma or equivalent

Bachelor's degree

Master's degree

- Terminal degree or higher
- 6. Do you have a managerial position, e.g. manager, supervisor?

Yes

No

For use by Amy Lynn Koehler-Warren only. Received from Mind Garden, Inc. on February 6, 2023

Multifactor Leadership Questionnaire Rater Form

Name of Leader:

Organization ID #:

Leader ID #:

Date:

This questionnaire is used to describe the leadership style of the above-mentioned individual as you perceive it. Answer all items on this answer sheet. If an item is irrelevant, or if you are unsure or do not know the answer, leave the answer blank. Please answer this questionnaire anonymously.

Important (necessary for processing): Which best describes you?

I am at a higher organizational level than the person I am rating.

The person I am rating is at my organizational level.

I am at a lower organizational level than the person I am rating.

Other than the above.

Forty-five descriptive statements are listed on the following pages. Judge how frequently each statement fits the person you are describing. Use the following rating scale:

Not at all	Once in a while	Sometimes	Fairly often	Frequently, if not always
0	1	2	3	4

The Person I Am Rating. . .

1.	Provides me with assistance in exchange for my efforts0	1	2	3	4
2.	*Re-examines critical assumptions to question whether they are appropriate0	1	2/	3	A
3.	Fails to interfere until problems become serious0	1/	2 /	3	4
4.	Focuses attention on irregularities, mistakes, exceptions, and deviations from standards	1	2	3	
5.	Avoids getting involved when important issues arise	1	2	¥	4
6.	*Talks about his/her most important values and beliefs	ł	2	3	4
7.	Is absent when needed	1	2	3	4
8.	*Seeks differing perspectives when solving problems	1	2	3	4
9.	*Talks optimistically about the future	1	2	3	4
10.	*Instills pride in me for being associated with him/her	1	2	3	4
11.	Discusses in specific terms who is responsible for achieving performance targets	1	2	3	4
12.	Waits for things to go wrong before taking action	1	2	3	4
13.	*Talks enthusiastically about what needs to be accomplished0	1	2	3	4
14.	*Specifies the importance of having a strong sense of purpose0	1	2	3	4
15.	*Spends time teaching and coaching0	1	2	3	4

Continued ->

116

Copyright 1995, 2000, 2004 by Bernard Bass and Bruce Avolio. All rights reserved. Published by Mind Garden, Inc., www.mindgarden.com

Not at all	Once in a while	Sometimes	Fairly often		Frec	uer t alv	ntly. vav	, S	
0	1	2	3			4		-	
Makes clear w	hat one can expect to r	eceive when performanc	e goals are achiev	ved	0	1	2	3	4
17. Shows that he	/she is a firm believer ir	n "If it ain't broke, don't fix	(it."		0	1	2	3	4
18. *Goes beyond	self-interest for the goo	od of the group			0	1	2	3	4
19. *Treats me as	an individual rather tha	n just as a member of a	group		0	1	2	3	4
20. Demonstrates	that problems must be	come chronic before taki	ng action		0	1	2	3	4
21. *Acts in ways t	hat builds my respect .				0	1	2	3	4
22. Concentrates h	his/her full attention on	dealing with mistakes, co	mplaints, and fail	ures	0	1	2	3	4
23. *Considers the	moral and ethical cons	sequences of decisions				1	2	3	4
24. Keeps track of	all mistakes				0	1/	2	_3	À
25. *Displays a ser	nse of power and confid	dence			0	1	Ł	Z	4
26. *Articulates a c	compelling vision of the	future			0	1	2	3	1
27. Directs my atte	ention toward failures to	meet standards			0	1	2	3	4
28. Avoids making	decisions		\neg		0	\mathbf{h}	2	3	4
29. *Considers me	as having different nee	eds, abilities, and aspirat	ions from others			1	2	3	4
30. *Gets me to lo	ok at problems from ma	any different angles			0	1	2	3	4
31. *Helps me to d	levelop my strengths				0	1	2	3	4
32. *Suggests new	v ways of looking at hoy	v to <i>e</i> omplete assignmen	ts		0	1	2	3	4
33. Delays respon	ding to urgent question	§ /			0	1	2	3	4
34. *Emphasizes t	he monortance of havin	g a collective sense of m	ission		0	1	2	3	4
35. Expresses sati	isfaction when I meete	xpectations			0	1	2	3	4
36. *Expresses co	nidence that goals will	be achieved			0	1	2	3	4
37. Is effective in n	neeting my job-related	needs			0	1	2	3	4
38. Uses methods	of leadership that are s	satisfying			0	1	2	3	4
39. Gets me to do	more than I expected t	o do			0	1	2	3	4
40. Is effective in r	epresenting me to high	er authority			0	1	2	3	4
41. Works with me	in a satisfactory way				0	1	2	3	4
42. Heightens my	desire to succeed				0	1	2	3	4
43. Is effective in n	meeting organizational	requirements			0	1	2	3	4
44. Increases my v	willingness to try harder	r			0	1	2	3	4
45. Leads a group	that is effective				0	1	2	3	4

For use by Amy Lynn Koehler-Warren only. Received from Mind Garden, Inc. on February 6, 2023

Copyright © 1995 by Bernard Bass and Bruce Avolio. All rights reserved. It is your legal responsibility to compensate the copyright holder of this work for any reproduction in any medium. If you need to reproduce the MLQ, please contact Mind Garden www.mindgarden.com. Mind Garden is a registered trademark of Mind Garden, Inc.

Copyright 1995, 2000, 2004 by Bernard Bass and Bruce Avolio. All rights reserved. Published by Mind Garden, Inc., www.mindgarden.com

NOSCAQ-50 Survey Instrument

In the following section please describe how you perceive that the managers and supervisors at this workplace deal with safety. Although some questions may appear very similar, please answer each one of them.

		Strongly disagree	Disagree	Agree	Strongly agree		
		Put only one X for each question					
1.	Management encourages employees here to work in accordance with safety rules - even when the work schedule is tight						
2.	Management ensures that everyone receives the necessary information on safety						
3.	Management looks the other way when someone is careless with safety						
4.	Management places safety before production						
5.	Management accepts employees here taking risks when the work schedule is tight						
6.	We who work here have confidence in the management's ability to deal with safety						
7.	Management ensures that safety problems discovered during safety rounds/evaluations are corrected immediately						
8.	When a risk is detected, management ignores it without action						
9.	Management lacks the ability to deal with safety properly						

		Put only one X for each question				
10.	Management strives to design safety routines that are meaningful and actually work					
11.	Management makes sure that everyone can influence safety in their work environment					
12.	Management encourages employees here to participate in decisions which affect their safety					
13.	Management never considers employees' suggestions regarding safety					
14.	Management strives for everybody at the worksite to have high competence concerning safety and risks					
15.	Management never asks employees for their opinions before making decisions regarding safety					
16.	Management involves employees in decisions regarding safety					
17.	Management collects accurate information in accident investigations					
18.	Fear of sanctions (negative consequences) from management discourages employees here from reporting near-miss accidents					
19.	Management listens carefully to all who have been involved in an accident					

Although some questions may appear very similar, please answer each one of them

		Strongly disagree Put or	Disagree	Agree	Strongly agree estion					
20.	Management looks for causes, not guilty persons, when an accident occurs									
21.	Management always blames employees for accidents									
22.	Management treats employees involved in an accident fairly									
In	In the following section please describe how you perceive that employees at this workplace deal with safety									
23.	We who work here try hard together to									

	achieve a high level of safety				
24.	We who work here take joint responsibility to ensure that the workplace is always kept tidy				
25.	We who work here do not care about each others' safety				
26.	We who work here avoid tackling risks that are discovered				
27.	We who work here help each other to work safely				
28.	We who work here take no responsibility for each others' safety				
		Strongly disagree	Strongly agree		
-----	---	----------------------	-------------------	---	--
		Put on	estion		
29.	We who work here regard risks as unavoidable				
30.	We who work here consider minor accidents to be a normal part of our daily work				
31.	We who work here accept dangerous behaviour as long as there are no accidents				
32.	We who work here break safety rules in order to complete work on time				
33.	We who work here never accept risk- taking even if the work schedule is tight				
34.	We who work here consider that our work is unsuitable for cowards				
35.	We who work here accept risk-taking at work				
			 	-	
36.	We who work here try to find a solution if someone points out a safety problem				
37.	We who work here feel safe when working together				
38.	We who work here have great trust in each others' ability to ensure safety				

Although some questions may appear very similar, please answer each one of them

	though some questions may appear very similar, please answer each one of them						
		Strongly disagree	Disagree	Agree	Strongly agree		
		Put only one X for each question					
39.	We who work here learn from our experiences in order to prevent accidents						
40.	We who work here take each others' opinions and suggestions concerning safety seriously						
41.	We who work here seldom talk about safety						
42.	We who work here always discuss safety issues when such issues come up						
43.	We who work here can talk freely and openly about safety						
44.	We who work here consider that a good safety representative plays an important role in preventing accidents						
45.	We who work here consider that safety rounds/evaluations have no effect on safety						
46.	We who work here consider that safety training to be good for preventing accidents						
47.	We who work here consider early planning for safety as meaningless						
48.	We who work here consider that safety rounds/evaluations help find serious hazards						
49.	We who work here consider safety training to be meaningless						
50.	We who work here consider it important to have clear-cut goals for safety						

....

JSS Survey Instrument

	JOB SATISFACTION SURVEY Paul E. Spector Department of Psychology University of South Florida Copyright Paul E. Spector 1994, All rights reserved.					
	PLEASE CIRCLE THE ONE NUMBER FOR EACH QUESTION THAT COMES CLOSEST TO REFLECTING YOUR OPINION ABOUT IT.	Disagree very much Disagree moderately Disagree slightly Agree slightly Agree moderately Agree very much				
1	I feel I am being paid a fair amount for the work I do.	1 2 3 4 5 6				
2	There is really too little chance for promotion on my job.	1 2 3 4 5 6				
3	My supervisor is quite competent in doing his/her job.	1 2 3 4 5 6				
4	I am not satisfied with the benefits I receive.	1 2 3 4 5 6				
5	When I do a good job, I receive the recognition for it that I should receive.	1 2 3 4 5 6				
6	Many of our rules and procedures make doing a good job difficult.	1 2 3 4 5 6				
7	I like the people I work with.	1 2 3 4 5 6				
8	I sometimes feel my job is meaningless.	1 2 3 4 5 6				
9	Communications seem good within this organization.	1 2 3 4 5 6				
10	Raises are too few and far between.	1 2 3 4 5 6				
11	Those who do well on the job stand a fair chance of being promoted.	1 2 3 4 5 6				
12	My supervisor is unfair to me.	1 2 3 4 5 6				
13	The benefits we receive are as good as most other organizations offer.	1 2 3 4 5 6				
14	I do not feel that the work I do is appreciated.	1 2 3 4 5 6				
15	My efforts to do a good job are seldom blocked by red tape.	1 2 3 4 5 6				
16	I find I have to work harder at my job because of the incompetence of people I work with.	1 2 3 4 5 6				
17	I like doing the things I do at work.	1 2 3 4 5 6				
18	The goals of this organization are not clear to me.	1 2 3 4 5 6				

	PLEASE CIRCLE THE ONE NUMBER FOR EACH QUESTION	Disagree very much									
	THAT COMES CLOSEST TO REFLECTING YOUR OPINION	Disagree moderately									
	ABOUT IT.	Disagree slightly									
	Copyright Paul E. Spector 1994, All rights reserved.			Agree slightly							
		Agree	e mo	derat	ely						
		Agree very much									
19	I feel unappreciated by the organization when I think about what they pay me.		1	2	3	4	5	6			
20	People get ahead as fast here as they do in other places.		1	2	3	4	5	6			
21	My supervisor shows too little interest in the feelings of subordinates.		1	2	3	4	5	6			
22	The benefit package we have is equitable.		1	2	3	4	5	6			
23	There are few rewards for those who work here.		1	2	3	4	5	6			
24	I have too much to do at work.		1	2	3	4	5	6			
25	I enjoy my coworkers.		1	2	3	4	5	6			
26	I often feel that I do not know what is going on with the organization.		1	2	3	4	5	6			
27	I feel a sense of pride in doing my job.		1	2	3	4	5	6			
28	I feel satisfied with my chances for salary increases.		1	2	3	4	5	6			
29	There are benefits we do not have which we should have.		1	2	3	4	5	6			
30	I like my supervisor.		1	2	3	4	5	6			
31	I have too much paperwork.		1	2	3	4	5	6			
32	I don't feel my efforts are rewarded the way they should be.		1	2	3	4	5	6			
33	I am satisfied with my chances for promotion.		1	2	3	4	5	6			
34	There is too much bickering and fighting at work.		1	2	3	4	5	6			
35	My job is enjoyable.		1	2	3	4	5	6			
36	Work assignments are not fully explained.		1	2	3	4	5	6	_		